Malay Kumar Kundu Durga Prasad Mohapatra Amit Konar Aruna Chakraborty Editors



Advanced Computing, Networking and Informatics – Volume 1

Advanced Computing and Informatics Proceedings of the Second International Conference on Advanced Computing, Networking and Informatics (ICACNI-2014)





Smart Innovation, Systems and Technologies

Volume 27

Series editors

Robert J. Howlett, KES International, Shoreham-by-Sea, UK e-mail: rjhowlett@kesinternational.org

Lakhmi C. Jain, University of Canberra, Canberra, Australia e-mail: Lakhmi.jain@unisa.edu.au

For further volumes: http://www.springer.com/series/8767

About this Series

The Smart Innovation, Systems and Technologies book series encompasses the topics of knowledge, intelligence, innovation and sustainability. The aim of the series is to make available a platform for the publication of books on all aspects of single and multi-disciplinary research on these themes in order to make the latest results available in a readily-accessible form. Volumes on interdisciplinary research combining two or more of these areas is particularly sought.

The series covers systems and paradigms that employ knowledge and intelligence in a broad sense. Its scope is systems having embedded knowledge and intelligence, which may be applied to the solution of world problems in industry, the environment and the community. It also focusses on the knowledge-transfer methodologies and innovation strategies employed to make this happen effectively. The combination of intelligent systems tools and a broad range of applications introduces a need for a synergy of disciplines from science, technology, business and the humanities. The series will include conference proceedings, edited collections, monographs, handbooks, reference books, and other relevant types of book in areas of science and technology where smart systems and technologies can offer innovative solutions.

High quality content is an essential feature for all book proposals accepted for the series. It is expected that editors of all accepted volumes will ensure that contributions are subjected to an appropriate level of reviewing process and adhere to KES quality principles.

Malay Kumar Kundu · Durga Prasad Mohapatra Amit Konar · Aruna Chakraborty Editors

Advanced Computing, Networking and Informatics – Volume 1

Advanced Computing and Informatics Proceedings of the Second International Conference on Advanced Computing, Networking and Informatics (ICACNI-2014)



Editors Malay Kumar Kundu Machine Intelligence Unit Indian Statistical Institute Kolkata India

Durga Prasad Mohapatra Dept. of Computer Science and Engineering National Institute of Technology Rourkela Rourkela India Amit Konar Dept. of Electronics and Tele-Communication Engineering Artificial Intelligence Laboratory Jadavpur University Kolkata India

Aruna Chakraborty Dept. of Computer Science and Engineering St. Thomas' College of Engineering & Technology Kidderpore India

 ISSN 2190-3018
 ISSN 2190-3026
 (electronic)

 ISBN 978-3-319-07352-1
 ISBN 978-3-319-07353-8
 (eBook)

 DOI 10.1007/978-3-319-07353-8
 springer Cham Heidelberg New York Dordrecht London
 (eBook)

Library of Congress Control Number: 2014940383

© Springer International Publishing Switzerland 2014

This work is subject to copyright. All rights are reserved by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed. Exempted from this legal reservation are brief excerpts in connection with reviews or scholarly analysis or material supplied specifically for the purpose of being entered and executed on a computer system, for exclusive use by the purchaser of the work. Duplication of this publication or parts thereof is permitted only under the provisions of the Copyright Law of the Publisher's location, in its current version, and permission for use must always be obtained from Springer. Permissions for use may be obtained through RightsLink at the Copyright Clearance Center. Violations are liable to prosecution under the respective Copyright Law.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

While the advice and information in this book are believed to be true and accurate at the date of publication, neither the authors nor the editors nor the publisher can accept any legal responsibility for any errors or omissions that may be made. The publisher makes no warranty, express or implied, with respect to the material contained herein.

Printed on acid-free paper

Springer is part of Springer Science+Business Media (www.springer.com)

Foreword

The present volume is an outcome, in the form of proceedings, of the 2nd International Conference on Advanced Computing, Networking and Informatics, St. Thomas' College of Engineering and Technology, Kolkata, India, June 24-26, 2014. As the name of the conference implies, the articles included herein cover a wide span of disciplines ranging, say, from pattern recognition, machine learning, image processing, data mining and knowledge discovery, soft computing, distributed computing, cloud computing, parallel and distributed networks, optical communication, wireless sensor networks, routing protocol and architecture to data privacy preserving, cryptology and data security, and internet computing. Each discipline, itself, has its own challenging problems and issues. Some of them are relatively more matured and advanced in theories with several proven application domains, while others fall in recent thrust areas. Interestingly, there are several articles, as expected, on symbiotic integration of more than one discipline, e.g., in designing intelligent networking and computing systems such as forest fire detection using wireless sensor network, minimizing call routing cost with assigned cell in wireless network, network intrusion detection system, determining load balancing strategy in cloud computing, and side lobe reduction and beam-width control, where the significance of pattern recognition, evolutionary strategy and soft computing has been demonstrated. This kind of interdisciplinary research is likely to grow significantly, and has strong promise in solving real life challenging problems.

The proceedings are logically split in two homogeneous volumes, namely, Advanced Computing and Informatics (vol. 1) and Wireless Networks and Security (vol. 2) with 81 and 67 articles respectively. The volumes fairly represent a state-of-the art of the research mostly being carried out in India in these domains, and are valued-additions to the current era of computing and knowledge mining.

The conference committee, editors, and the publisher deserve congratulations for organizing the event (ICACNI-2014) which is very timely, and bringing out the archival volumes nicely as its output.

Kolkata, April 2014

Sankar K. Pal Distinguished Scientist and former Director Indian Statistical Institute

Message from the Honorary General Chair

It gives me great pleasure to introduce the *International Conference on Advanced Computing, Networking and Informatics (ICACNI 2014)* which will be held at St. Thomas' College of Engineering and Technology, Kolkata during June 24–26, 2014. ICACNI is just going to cross its second year, and during this small interval of time it has attracted a large audience. The conference received over 650 submissions of which only 148 papers have been accepted for presentation. I am glad to note that ICACNI involved top researchers from 26 different countries as advisory board members, program committee members and reviewers. It also received papers from 10 different countries.

ICACNI offers an interesting forum for researchers of three apparently diverse disciplines: Advanced Computing, Networking and Informatics, and attempts to focus on engineering applications, covering security, cognitive radio, human-computer interfacing among many others that greatly rely on these cross-disciplinary research outcomes. The accepted papers are categorized into two volumes, of which volume 1 includes all papers on advanced computing and informatics, while volume 2 includes accepted papers on wireless network and security. The volumes will be published by Springer-Verlag.

The conference includes plenary lecture, key-note address and four invited sessions by eminent scientists from top Indian and foreign research/academic institutes. The lectures by these eminent scientists will provide an ideal platform for dissemination of knowledge among researchers, students and practitioners. I take this opportunity to thank all the participants, including the keynote, plenary and invited speakers, reviewers, and the members of different committees in making the event a grand success. Thanks are also due to the various Universities/Institutes for their active support towards this endeavor, and lastly Springer-Verlag for publishing the proceedings under their prestigious *Smart Innovation, Systems and Technologies (SIST) series.*

Wish the participants an enjoyable and productive stay in Kolkata.

(X. Lithayunds

Kolkata, April 2014

Dwijesh Dutta Majumder Honorary General Chair ICACNI -2014

Preface

The twenty first century has witnessed a paradigm shift in three major disciplines of knowledge: 1) Advanced/Innovative computing ii) Networking and wireless Communications and iii) informatics. While the first two are complete in themselves by their titles, the last one covers several sub-disciplines involving geo-, bio-, medical and cognitive informatics among many others. Apparently, the above three disciplines of knowledge are complementary and mutually exclusive but their convergence is observed in many real world applications, encompassing cyber-security, internet banking, health-care, sensor networks, cognitive radio, pervasive computing and many others.

The International Conference on *Advanced Computing, Networking and Informatics* (ICACNI) is aimed at examining the convergence of the above three modern disciplines through interactions among three groups of people. The first group comprises leading international researchers, who have established themselves in one of the above three thrust areas. The plenary, the keynote lecture and the invited talks are organized to disseminate the knowledge of these academic experts among young researchers/practitioners of the respective domain. The invited talks are also expected to inspire young researchers to initiate/orient their research in respective fields. The second group of people comprises Ph.D./research students, working in the cross-disciplinary areas, who might be benefited from the first group and at the same time may help creating interest in the cross-disciplinary research areas among the academic community, including young teachers and practitioners. Lastly, the group comprising undergraduate and master students would be able to test the feasibility of their research through feedback of their oral presentations.

ICACNI is just passing its second birthday. Since its inception, it has attracted a wide audience. This year, for example, the program committee of ICACNI received as many as 646 papers. The acceptance rate is intentionally kept very low to ensure a quality publication by Springer. This year, the program committee accepted only 148 papers from these 646 submitted papers. An accepted paper has essentially received very good recommendation by at least two experts in the respective field.

To maintain a high standard of ICACNI, researchers from top international research laboratories/universities have been included in both the advisory committee and the program committee. The presence of these great personalities has helped the conference to develop its publicity during its infancy and promote it quality through an academic exchange among top researchers and scientific communities.

The conference includes one plenary session, one keynote address and four invited speech sessions. It also includes 3 special sessions and 21 general sessions (altogether 24 sessions) with a structure of 4 parallel sessions over 3 days. To maintain good question-answering sessions and highlight new research results arriving from the sessions, we selected subject experts from specialized domains as session chairs for the conference. ICACNI also involved several persons to nicely organize registration, take care of finance, hospitality of the authors/audience and other supports. To have a closer interaction among the people of the organizing committee, all members of the organizing committee have been selected from St. Thomas' College of Engineering and Technology.

The papers that passed the screening process by at least two reviewers, wellformatted and nicely organized have been considered for publication in the Smart Innovations Systems Technology (SIST) series of Springer. The hard copy proceedings include two volumes, where the first volume is named as *Advanced Computing and Informatics* and the second volume is named as *Wireless Networks and Security*. The two volumes together contain 148 papers of around eight pages each (in Springer LNCS format) and thus the proceedings is expected to have an approximate length of 1184 pages.

The editors gratefully acknowledge the contribution of the authors and the entire program committee without whose active support the proceedings could hardly attain the present standards. They would like to thank the keynote speaker, the plenary speaker, the invited speakers and also the invited session chairs, the organizing chair along with the organizing committee and other delegates for extending their support in various forms to ICACNI-2014. The editors express their deep gratitude to the Honorary General Chair, the General Chair, the Advisory Chair and the Advisory board members for their help and support to ICACNI-2014. The editors are obliged to Prof. Lakhmi C. Jain, the academic series editor of the SIST series, Springer and Dr. Thomas Ditzinger, Senior Editor, Springer, Heidelberg for extending their co-operation in publishing the proceeding in the prestigious SIST series of Springer. They also like to mention the hard efforts of Mr. Indranil Dutta of the Machine Intelligence Unit of ISI Kolkata for the editorial support. The editors also acknowledge the technical support they received from the students of ISI, Kolkata and Jadavpur University and also the faculty of NIT Rourkela and St. Thomas' College of Engineering and Technology without which the work could not be completed in right time. Lastly, the editors thank Dr. Sailesh Mukhopadhyay, Prof. Gautam Banerjee and Dr. Subir Chowdhury of St. Thomas' College of Engineering and Technology for their support all the way long to make this conference a success.

Kolkata April 14, 2014 Malay Kumar Kundu Durga Prasad Mohapatra Amit Konar Aruna Chakraborty

Organization

Advisory Chair

Sankar K. Pal

Distinguished Scientist, Indian Statistical Institute, India

Nazarbayev University, Republic of Kazakhstan

Advisory Board Members

Alex P. James Anil K. Kaushik Atilla Elci Atulya K. Nagar Brijesh Verma Debajyoti Mukhopadhyay Debatosh Guha George A. Tsihrintzis Hugo Proenca Jocelyn Chanussot Kenji Suzuki Khalid Saeed Klaus David Maode Ma Massimiliano Rak Massimo Tistarelli Nishchal K. Verma Pascal Lorenz Phalguni Gupta Prasant Mohapatra Prasenjit Mitra Raj Jain

Rajesh Siddavatam

Additional Director, DeitY, Govt. of India, India Aksaray University, Turkey Liverpool Hope University, United Kingdom Central Queensland University, Australia Maharashtra Institute of Technology, India University of Calcutta, India University of Piraeus, Greece University of Beira Interior, Portugal Grenoble Institute of Technology, France The University of Chicago, Chicago AGH University of Science and Technology, Poland University of Kassel, Germany Nanyang Technological University, Singapore Second University of Naples, Italy University of Sassari, Italy Indian Institute of Technology, Kanpur, India University of Haute Alsace, France Indian Institute of Technology Kanpur, India University of California, USA Pennsylvania State University, USA Washington University in St. Louis, USA Kalinga Institute of Industrial Technology, India

Rajkumar Buyya	The University of Melbourne, Australia
Raouf Boutaba	University of Waterloo, Canada
Sagar Naik	University of Waterloo, Canada
Salvatore Vitabile	University of Palermo, Italy
Sansanee Auephanwiriyakul	Chiang Mai University, Thailand
Subhash Saini	The National Aeronautics and Space
	Administration (NASA), USA

ICACNI Conference Committee

Chief Patron

Sailesh Mukhopadhyay St. Thomas' College of Engineering and Technology, Kolkata, India Patron Gautam Banerjea St. Thomas' College of Engineering and Technology, Kolkata, India **Honorary General Chair** Dwijesh Dutta Majumder Professor Emeritus, Indian Statistical Institute, Kolkata, India Institute of Cybernetics Systems and Information Technology, India Director, Governing Board, World Organization of Systems and Cybernetics (WOSC), Paris **General Chairs** Rajib Sarkar Central Institute of Technology Raipur, India Mrithunjoy Bhattacharyya St. Thomas' College of Engineering and Technology, Kolkata, India **Programme Chairs** Malay Kumar Kundu Indian Statistical Institute, Kolkata, India Amit Konar Jadavpur University, Kolkata, India St. Thomas' College of Engineering and Aruna Chakraborty Technology, Kolkata, India **Programme Co-chairs** Asit Kumar Das Bengal Engineering and Science University, Kolkata, India Ramjeevan Singh Thakur Maulana Azad National Institute of Technology, India Umesh A. Deshpande Sardar Vallabhbhai National Institute of Technology, India

Organizing Chairs

Ashok K. Turuk	National Institute of Technology Rourkela,
	Rourkela, India
Rabindranath Ghosh	St. Thomas' College of Engineering and
	Technology, Kolkata, India

Technical Track Chairs

Joydeb Roychowdhury	Central Mechanical Engineering Research Institute,
	India
Korra Sathyababu	National Institute of Technology Rourkela, India
Manmath Narayan Sahoo	National Institute of Technology Rourkela, India

Honorary Industrial Chair

G.C. Deka	Ministry of Labour & Employment, Government of
	India

Industrial Track Chairs

Umesh Chandra Pati	National Institute of Technology Rourkela, India
Bibhudutta Sahoo	National Institute of Technology Rourkela, India

Special Session Chairs

Ashish Agarwal	Boston University, USA
Asutosh Kar	International Institute of Information Technology
	Bhubaneswar, India
Daya K. Lobiyal	Jawaharlal Nehru University, India
Mahesh Chandra	Birla Institute of Technology, Mesra, India
Mita Nasipuri	Jadavpur University, Kolkata, India
Nandini Mukherjee	Chairman, IEEE Computer Society, Kolkata Chapter
	Jadavpur University, Kolkata, India
Ram Shringar Rao	Ambedkar Institute of Advanced Communication Technologies & Research, India
Web Chair	
Indranil Dutta	Indian Statistical Institute, Kolkata, India
Publication Chair	
Sambit Bakshi	National Institute of Technology Rourkela, India

Publicity Chair

Mohammad Ayoub Khan Center for Development of Advanced Computing, India

Organizing Committee

Amit Kr. Siromoni Anindita Ganguly

Arindam Chakravorty

Dipak Kumar Kole

Prasanta Kumar Sen

Ramanath Datta

Subarna Bhattacharya

Supriya Sengupta

Program Committee

Vinay. A

Chunyu Ai

Rashid Ali C.M. Ananda Soumen Bag

Sanghamitra Bandyopadhyay Punam Bedi Dinabandhu Bhandari Paritosh Bhattacharya Malay Bhattacharyya Sambhunath Biswas Darko Brodic Sasthi C. Ghosh R.C. Hansdah Nabendu Chaki Goutam Chakraborty St. Thomas' College of Engineering and Technology, Kolkata, India

- St. Thomas' College of Engineering and Technology, Kolkata, India
- St. Thomas' College of Engineering and Technology, Kolkata, India
- St. Thomas' College of Engineering and Technology, Kolkata, India
- St. Thomas' College of Engineering and Technology, Kolkata, India
- St. Thomas' College of Engineering and Technology, Kolkata, India
- St. Thomas' College of Engineering and Technology, Kolkata, India
- St. Thomas' College of Engineering and Technology, Kolkata, India

Peoples Education Society Institute of Technology, Bangalore, India University of South Carolina Upstate, Spartanburg, USA Aligarh Muslim University, Aligarh, India National Aerospace Laboratories, Bangalore, India International Institution of Information Technology, Bhubaneswar, India Indian Statistical Institute, Kolkata, India University of Delhi, Delhi, India Heritage Institute of Technology, Kolkata, India National Institute of Technology, Agartala, India University of Kalyani, Kolkata, India Indian Statistical Institute, Kolkata, India University of Belgrade, Bor, Serbia Indian Statistical Institute, Kolkata, India Indian Institute of Science, Bangalore, India University of Calcutta, Kolkata, India Iwate Prefectural University, Takizawa, Japan

Mihir Chakraborty Amita Chatterjee Debasis Chaudhuri Lopamudra Chowdhury Nabanita Das Soumya Kanti Datta Rajat De Utpal Garain Ashis Ghosh Mukesh Goswami Yogesh H. Dandawate

Biju Issac Lakhmi C. Jain R. Janarthanan Boleslaw K. Szymanski Tienfuan Kerh

Zahoor Khan Dakshina Ranjan Kisku Sotiris Kotsiantis Dipak Kr. Kole

Aswani Kumar Cherukuri Swapan Kumar Parui Prasanta Kumar Pradhan

B. Narendra Kumar Rao

Flavio Lombardi Pradipta Maji Raghvendra Mall Amiya Kr. Mallick

Debaprasad Mandal Ujjwal Maulik Pabitra Mitra Imon Mukherjee

Nandini Mukherjee Jayanta Mukhopadhyay C.A. Murthy M. Murugappan Mita Nasipuri Rajdeep Niyogi Indian Statistical Institute, Kolkata, India Jadavpur University, Kolkata, India Indian Institute of Management, Kolkata, India Jadavpur University, Kolkata, India Indian Statistical Institute, Kolkata, India Institute Eurecom, Sophia Antipolis, France Indian Statistical Institute, Kolkata, India Indian Statistical Institute, Kolkata, India Indian Statistical Institute, Kolkata, India Dharmsinh Desai University, Gugrat, India Vishwakarma Institute of Information Technology, Pune. India Teesside University, Middlesbrough, UK University of South Australia, Adelaide, Australia T. J. S. Engineering College, Chennai, India Rensselaer Polytechnic Institute, New York, USA National Pingtung University of Science and Technology, Pingtung, Taiwan Dalhouise University, Halifax, Canada Asansole Engineering College, Asansole, India University of Patras, Hellas, Greece St. Thomas' College of Engineering and Technology, Kolkata, India VIT University, Vellore, India Indian Statistical Institute, Kolkata, India St. Thomas' College fo Engineering and Technology, Kolkata, India Sree Vidyanikethan Engineering College, Tirupati, India Roma Tre University of Rome, Rome, Italy Indian Statistical Institute, Kolkata, India University of Leuven, Leuven, Belgium St. Thomas' College fo Engineering and Technology, Kolkata, India Indian Statistical Institute, Kolkata, India Jadavpur University, Kolkata, India Indian Institute of Technology, Kharagpur, India St. Thomas' College fo Engineering and Technology, Kolkata, India Jadavpur University, Kolkata, India Indian Institute of Technology, Kharagpur, India Indian Statistical Institute, Kolkata, India University of Malayesia, Malayesia Jadavpur University, Kolkata, India Indian Institute of Technology, Roorkee, India

Steven Noel M.C. Padma Rajarshi Pal

Umapada Pal Anika Pflug

Surya Prakash Ganapatsingh G. Rajput

Anca Ralescu Umesh Hodeghatta Rao

Ajay K. Ray

Tuhina Samanta

Andrey V. Savchenko

Bimal Bhusan Sen

Indranil Sengupta Patrick Siarry Nanhay Singh

Pradeep Singh Vivek Singh Bhabani P. Sinha Sundaram Suresh Jorge Sá Silva Vasile Teodor Dadarlat

B. Uma Shankar M. Umaparvathi Palaniandavar Venkateswaran Stefan Weber Azadeh Ghandehari Ch. Aswani Kumar Cristinel Ababei Dilip Singh Sisodia Jamuna Kanta Sing Krishnan Nallaperumal Manu Pratap Singh Narayan C. Debnath Naveen Kumar George Manson University, Fairfax, USA PES College of Engineering, Karnataka, India Institute for Development and Research in Banking Technology, Hyderabad, India Indian Statistical Institute, Kolkata, India Hochschule Darmstadt - CASED, Darmstadt, Germany Indian Institute of Technology, Indore, India Rani Channamma University, Karnataka, India University of Cincinnati, Ohio, USA Xavier Institute of Management, Bhubaneswar, India Bengal Engineering and Science University, Shibpur, India Bengal Engineering and Science University, Shibpur, India National Research University Higher School of Economics, Molscow, Russia St. Thomas' College fo Engineering and Technology, Kolkata, India Indian Institute of Technology, Kharagpur, India Universite de Paris, Paris Ambedkar Institute of Advanced Communication Technologies & Research, Delhi, India National Institute of Technology, Raipur, India South Asian University, New Delhi, India Indian Statistical Institute, Kolkata, India Nanyang Technological University, Singapore University of Coimbra, Portugal Technical University of Cluj Napoca, Cluj Napoca, Romania Indian Statistical Institute, Kolkata, India RVS College of Engineering, Coimbatore, India Jadavpur University, Kolkata, India Trinity College, Dublin, Ireland Islamic Azad University, Tehran, Iran Vellore Institute of Technology, India University at Buffalo, USA National Institute of Technology Raipur, India Jadavpur University, Kolkata, India Sundaranar University, India Dr. B.R. Ambedkar University, Agra, India Winona State University, USA Indira Gandhi National Open University, India

National Institute of Technology Silchar, India

Delhi Technological University, India

Technology, Kolkata, India The Pennsylvania State University, USA

Jawaharlal Nehru University, India

St. Thomas' College of Engineering and

Indian Institute of Information Technology,

Galgotias University, India

Gwalior. India

Nidul Sinha Sanjay Kumar Soni Sanjoy Das Subir Chowdhury

Syed Rizvi Sushil Kumar Anupam Sukhla

Additional Reviewers

A.M., Chandrashekhar Acharva, Anal Agarwal, Shalabh B.S., Mahanand Bandyopadhyay, Oishila Barpanda, Soubhagya Sankar Basu, Srinka Battula, Ramesh Babu Bhattacharjee, Sourodeep Bhattacharjee, Subarna Bhattacharya, Indrajit Bhattacharya, Nilanjana Bhattacharyya, Saugat Bhowmik, Deepayan Biswal, Pradyut Biswas, Rajib Bose, Subrata Chakrabarti, Prasun Chakraborty, Debashis Chakraborty, Jayasree Chandra. Helen Chandra. Mahesh Chatterjee, Aditi Chatterjee, Sujoy Chowdhury, Archana Chowdhury, Manish Dalai, Asish Darbari, Manuj Das, Asit Kumar Das, Debaprasad Das. Nachiketa Das, Sudeb Datta, Biswajita

Datta, Shreyasi De. Debashis Dhabal, Supriya Dhara, Bibhas Chandra Duvvuru, Rajesh Gaidhane, Vilas Ganguly, Anindita Garg, Akhil Ghosh Dastidar, Jayati Ghosh. Arka Ghosh. Lidia Ghosh, Madhumala Ghosh, Partha Ghosh, Rabindranath Ghosh, Soumyadeep Ghoshal, Ranjit Goyal, Lalit Gupta, Partha Sarathi Gupta, Savita Halder, Amiya Halder. Santanu Herrera Lara. Roberto Jaganathan, Ramkumar Kakarla, Jagadeesh Kar. Mahesh Kar. Reshma Khasnobish, Anwesha Kole, Dipak Kumar Kule, Malay Kumar, Raghvendra Lanka. Swathi Maruthi, Padmaja Mishra, Dheerendra Mishra, Manu

Misra, Anuranjan Mohanty, Ram Maitra, Subhamoy Mondal, Jaydeb Mondal, Tapabrata Mukherjee, Nabanita Mukhopadhyay, Debajyoti Mukhopadhyay, Debapriyay Munir, Kashif Nasim Hazarika. Saharriyar Zia Nasipuri, Mita Neogy, Sarmistha Pal, Monalisa Pal. Tamaltaru Palodhi, Kanik Panigrahi, Ranjit Pati, Soumen Kumar Patil, Hemprasad Patra, Braja Gopal Pattanayak, Sandhya Paul. Amit Paul, Partha Sarathi Phadikar, Amit Phadikar. Santanu Poddar, Soumyajit Prakash. Neerai Rakshit, Pratyusha Raman, Rahul Ray, Sumanta Roy, Pranab Roy, Souvik Roy, Swapnoneel

Rup, Suvendu Sadhu, Arup Kumar Saha Ray, Sanchita Saha, Anuradha Saha, Anushri Saha, Dibakar Saha, Indrajit Saha, Sriparna Sahoo, Manmath N. Sahu, Beeren Sanyal, Atri Sardar, Abdur Sarkar, Apurba Sarkar, Dhrubasish Sarkar, Ushnish Sen, Sonali Sen, Soumya Sethi, Geetika Sharma, Anuj Tomar, Namrata Umapathy, Latha Upadhyay, Anjana Wankhade, Kapil

Contents

Signal and Speech Analysis

Time-Domain Solution of Transmission through Multi-modeled Obstacles for UWB Signals Sanjay Soni, Bajrang Bansal, Ram Shringar Rao Indexing and Retrieval of Speech Documents 1 Piyush Kumar P. Singh, K.E. Manjunath, R. Ravi Kiran, 1 Jainath Yadav, K. Sreenivasa Rao 1 An Improved Filtered-x Least Mean Square Algorithm for Acoustic 2 Noise Suppression 2 Asutosh Kar, Ambika Prasad Chanda, Sarthak Mohapatra, 3 Mahesh Chandra 3 Pranab Das, Abhishek Deb, Asutosh Kar, Mahesh Chandra 3 On the Dissimilarity of Orthogonal Least Squares and Orthogonal 4 Arvinder Kaur, Sumit Budhiraja 4	1
Indexing and Retrieval of Speech Documents 1 Piyush Kumar P. Singh, K.E. Manjunath, R. Ravi Kiran, 1 Jainath Yadav, K. Sreenivasa Rao 1 An Improved Filtered-x Least Mean Square Algorithm for Acoustic 1 Noise Suppression 2 Asutosh Kar, Ambika Prasad Chanda, Sarthak Mohapatra, 2 Mahesh Chandra 3 A Unique Low Complexity Parameter Independent Adaptive Design for 3 Pranab Das, Abhishek Deb, Asutosh Kar, Mahesh Chandra 3 On the Dissimilarity of Orthogonal Least Squares and Orthogonal 4 Arvinder Kaur, Sumit Budhiraja 4	9
An Improved Filtered-x Least Mean Square Algorithm for Acoustic 2 Noise Suppression 2 Asutosh Kar, Ambika Prasad Chanda, Sarthak Mohapatra, 2 Mahesh Chandra 3 A Unique Low Complexity Parameter Independent Adaptive Design for 3 Echo Reduction 3 Pranab Das, Abhishek Deb, Asutosh Kar, Mahesh Chandra 3 On the Dissimilarity of Orthogonal Least Squares and Orthogonal 4 Arvinder Kaur, Sumit Budhiraja 4	17
A Unique Low Complexity Parameter Independent Adaptive Design for 3 Echo Reduction 3 Pranab Das, Abhishek Deb, Asutosh Kar, Mahesh Chandra 3 On the Dissimilarity of Orthogonal Least Squares and Orthogonal 4 Matching Pursuit Compressive Sensing Reconstruction 4 Arvinder Kaur, Sumit Budhiraja 4	25
On the Dissimilarity of Orthogonal Least Squares and Orthogonal Matching Pursuit Compressive Sensing Reconstruction 4 Arvinder Kaur, Sumit Budhiraja	33
	41
Machine Learning	
Machine Learning Based Shape Classification Using Tactile Sensor Array 4 Dennis Babu, Sourodeep Bhattachariee, Irin Bandvopadhvava.	47

Joydeb Roychowdhury

Multi-view Ensemble Learning for Poem Data Classification Using SentiWordNet	57
Vipin Kumar, Sonajharia Minz	
A Prototype of an Intelligent Search Engine Using Machine Learning Based Training for Learning to Rank Piyush Rai, Shrimai Prabhumoye, Pranay Khattri, Love Rose Singh Sandhu, S. Sowmya Kamath	67
Vegetable Grading Using Tactile Sensing and Machine Learning <i>Irin Bandyopadhyaya, Dennis Babu, Sourodeep Bhattacharjee,</i> <i>Joydeb Roychowdhury</i>	77
Neural Networks with Online Sequential Learning Ability for a Reinforcement Learning Algorithm Hitesh Shah, Madan Gopal	87
Pattern Analysis and Recognition	
Scatter Matrix versus the Proposed Distance Matrix on Linear Discriminant Analysis for Image Pattern Recognition E.S. Gopi, P. Palanisamy	101
Gender Recognition Using Fusion of Spatial and Temporal Features Suparna Biswas, Jaya Sil	109
The Use of Artificial Intelligence Tools in the Detection of Cancer Cervix Lamia Guesmi, Omelkir Boughzala, Lotfi Nabli, Mohamed Hedi Bedoui	117
A Scalable Feature Selection Algorithm for Large Datasets – Quick Branch and Bound Iterative (QBB-I) Prema Nedungadi, M.S. Remya	125
Towards a Scalable Approach for Mining Frequent Patterns from theLinked Open Data CloudRajesh Mahule, O.P. Vyas	137
Automatic Synthesis of Notes Based on Carnatic Music Raga Characteristics Janani Varadharajan, Guruprasad Sridharan, Vignesh Natarajan, Rajeswari Sridhar	145
Smart Card Application for Attendance Management System Shalini Jain, Anupam Shukla	153
Performance Evaluation of GMM and SVM for Recognition of Hierarchical Clustering Character	161

Data Clustering and Zonation of Earthquake Building Damage HazardArea Using FKCN and Kriging AlgorithmEdy Irwansyah, Sri Hartati	171
Parametric Representation of Paragraphs and Their Classification Dinabandhu Bhandari, Partha Sarathi Ghosh	179
LDA Based Emotion Recognition from Lyrics	187
Efficient Approach for Near Duplicate Document Detection Using Textual and Conceptual Based Techniques Rajendra Kumar Roul, Sahil Mittal, Pravin Joshi	195
Decision Tree Techniques Applied on NSL-KDD Data and Its Comparison with Various Feature Selection Techniques H.S. Hota, Akhilesh Kumar Shrivas	205
Matra and Tempo Detection for INDIC Tala -s Susmita Bhaduri, Sanjoy Kumar Saha, Chandan Mazumdar	213
Image Analysis	
Modified Majority Voting Algorithm towards Creating Reference Image for Binarization Ayan Dey, Soharab Hossain Shaikh, Khalid Saeed, Nabendu Chaki	221
Multiple People Tracking Using Moment Based Approach	229
Wavelets-Based Clustering Techniques for Efficient Color ImageSegmentationParitosh Bhattacharya, Ankur Biswas, Santi Prasad Maity	237
An Approach of Optimizing Singular Value of YCbCr Color Space with q-Gaussian Function in Image Processing Abhisek Paul, Paritosh Bhattacharya, Santi Prasad Maity	245
Motion Tracking of Humans under Occlusion Using Blobs	251
Efficient Lifting Scheme Based Super Resolution Image Reconstruction Using Low Resolution Images	259
Improved Chan-Vese Image Segmentation Model Using Delta-Bar-Delta Algorithm Devraj Mandal, Amitava Chatterjee, Madhubanti Maitra	267

Online Template Matching Using Fuzzy Moment Descriptor Arup Kumar Sadhu, Pratyusha Das, Amit Konar, Ramadoss Janarthanan	275
Classification of High Resolution Satellite Images Using Equivariant Robust Independent Component Analysis Pankaj Pratap Singh, R.D. Garg	283
3D Face Recognition across Pose Extremities	291
Indexing Video Database for a CBVCD System Debabrata Dutta, Sanjoy Kumar Saha, Bhabatosh Chanda	301
Fuzzy Set Theoretic Analysis	
Data Dependencies and Normalization of Intuitionistic Fuzzy Databases Asma R. Shora, Afshar Alam	309
Fuzzy Logic Based Implementation for Forest Fire Detection UsingWireless Sensor NetworkMamata Dutta, Suman Bhowmik, Chandan Giri	319
Fuzzy Connectedness Based Segmentation of Fetal Heart from ClinicalUltrasound ImagesSridevi Sampath, Nirmala Sivaraj	329
An Improved Firefly Fuzzy C-Means (FAFCM) Algorithm for Clustering Real World Data Sets Janmenjoy Nayak, Matrupallab Nanda, Kamlesh Nayak, Bighnaraj Naik, Himansu Sekhar Behera	339
On Kernel Based Rough Intuitionistic Fuzzy C-means Algorithm and a Comparative Analysis B.K. Tripathy, Anurag Tripathy, K. Govindarajulu, Rohan Bhargav	349
FuSCa: A New Weighted Membership Driven Fuzzy Supervised Classifier . <i>Pritam Das, S. SivaSathya, K. Joshil Raj</i>	361
Choice of Implication Functions to Reduce Uncertainty in Interval Type-2 Fuzzy Inferences	369
Detection of Downy Mildew Disease Present in the Grape Leaves Based on Fuzzy Set Theory Dipak Kumar Kole, Arya Ghosh, Soumya Mitra	377

Facial Expression Synthesis for a Desired Degree of Emotion Using Fuzzy Abduction	385
Sumantra Chakraborty, Sudipta Ghosh, Amit Konar, Saswata Das, Ramadoss Janarthanan	
Document Analysis	
A Novel Semantic Similarity Based Technique for Computer Assisted Automatic Evaluation of Textual Answers Udit Kr. Chakraborty, Samir Roy, Sankhayan Choudhury	393
Representative Based Document Clustering	403
A New Parallel Thinning Algorithm with Stroke Correction for Odia	
Characters	413
Evaluation of Collaborative Filtering Based on Tagging with DiffusionSimilarity Using Gradual Decay ApproachLatha Banda, Kamal Kanth Bharadwaj	421
Rule Based Schwa Deletion Algorithm for Text to Speech Synthesis inHindiShikha Kabra, Ritika Agarwal, Neha Yadav	429
Unsupervised Word Sense Disambiguation for Automatic Essay Scoring <i>Prema Nedungadi, Harsha Raj</i>	437
Biometric and Biological Data Analysis	
Disjoint Tree Based Clustering and Merging for Brain Tumor Extraction <i>Ankit Vidyarthi, Namita Mittal</i>	445
Segmentation of Acute Brain Stroke from MRI of Brain Image UsingPower Law Transformation with Accuracy EstimationSudipta Roy, Kingshuk Chatterjee, Samir Kumar Bandyopadhyay	453
A Hough Transform Based Feature Extraction Algorithm for Finger Knuckle Biometric Recognition System Usha Kazhagamani, Ezhilarasan Murugasen	463
An Efficient Multiple Classifier Based on Fast RBFN for Biometric	470
Sumana Kundu, Goutam Sarker	4/3
Automatic Tortuosity Detection and Measurement of Retinal Blood Vessel	107
Sk. Latib, Madhumita Mukherjee, Dipak Kumar Kole, Chandan Giri	400

A New Indexing Method for Biometric Databases Using Match Scores and Decision Level Fusion	493
Data and Web Mining	
Split-Encoding: The Next Frontier Tool for Big Data	501
Identification of Lost or Deserted Written Texts Using Zipf's Law with NLTK Devanshi Gupta, Priyank Singh Hada, Deepankar Mitra, Niket Sharma	511
An Efficient Approach for Discovering Closed Frequent Patterns in High Dimensional Data Sets Bharat Singh, Raghvendra Singh, Nidhi Kushwaha, O.P. Vyas	519
Time-Fading Based High Utility Pattern Mining from Uncertain Data Streams Chiranjeevi Manike, Hari Om	529
Classification for Multi-Relational Data Mining Using Bayesian Belief Network Nileshkumar D. Bharwad, Mukesh M. Goswami	537
Noise Elimination from Web Page Based on Regular Expressions for Web Content Mining Amit Dutta, Sudipta Paria, Tanmoy Golui, Dipak Kumar Kole	545
e-Learning and e-Commerece	
Modified Literature Based Approach to Identify Learning Styles in Adaptive E-Learning Sucheta V. Kolekar, Radhika M. Pai, M.M. Manohara Pai	555
A Concept Map Approach to Supporting Diagnostic and Remedial Learning Activities Anal Acharya, Devadatta Sinha	565
Data Prediction Based on User Preference Deepak Kumar, Gopalji Varshney, Manoj Thakur	575
Ontological Analysis	
Automatic Resolution of Semantic Heterogeneity in GIS:An Ontology Based ApproachShrutilipi Bhattacharjee, Soumya K. Ghosh	585

Web-Page Indexing Based on the Prioritized Ontology TermsSukanta Sinha, Rana Dattagupta, Debajyoti Mukhopadhyay		
A Hybrid Approach Using Ontology Similarity and Fuzzy Logic for Semantic Question Answering Monika Rani, Maybin K. Muyeba, O.P. Vyas	601	
Ontology Based Object-Attribute-Value Information Extraction from Web Pages in Search Engine Result Retrieval	611	
V. Vijayarajan, M. Dinakaran, Mayank Lohani		
Human-Computer Interfacing		
Effects of Robotic Blinking Behavior for Making Eye Contact with	(01	
Humans Mohammed Moshiul Hoque, Quazi Delwar Hossian, Kaushik Deb	621	
Improvement and Estimation of Intensity of Facial ExpressionRecognition for Human-Computer InteractionKunal Chanda, Washef Ahmed, Soma Mitra, Debasis Mazumdar	629	
Cognitive Activity Recognition Based on Electrooculogram Analysis Anwesha Banerjee, Shreyasi Datta, Amit Konar, D.N. Tibarewala, Janarthanan Ramadoss	637	
Detection of Fast and Slow Hand Movements from Motor Imagery EEGSignalsSaugat Bhattacharyya, Munshi Asif Hossain, Amit Konar,D.N. Tibarewala, Janarthanan Ramadoss	645	
Swarm and Evolutionary Computing		
A Solution of Degree Constrained Spanning Tree Using Hybrid GA with Directed Mutation Sounak Sadhukhan, Samar Sen Sarma	653	
Side Lobe Reduction and Beamwidth Control of Amplitude Taper BeamSteered Linear Array Using Tschebyscheff Polynomial and ParticleSwarm OptimizationPrarthana Mukherjee, Ankita Hajra, Sauro Ghosal, Soumyo Chatterjee,Sayan Chatterjee	661	
Pervasive Diary in Music Rhythm Education: A Context-Aware Learning Tool Using Genetic Algorithm <i>Sudipta Chakrabarty, Samarjit Roy, Debashis De</i>	669	

An Elitist Binary PSO Algorithm for Selecting Features in High Dimensional Data Suresh Dara, Haider Banka	679
An Immune System Inspired Algorithm for Protein Function Prediction Archana Chowdhury, Amit Konar, Pratyusha Rakshit, Ramadoss Janarthanan	687
Fast Approximate Eyelid Detection for PeriocularLocalizationSaharriyar Zia Nasim Hazarika, Neeraj Prakash, Sambit Bakshi,Rahul Raman	697
Prediction of an Optimum Parametric Combination for Minimum Thrust Force in Bone Drilling: A Simulated Annealing Approach	705
Author Index	715

Application of Bilinear Recursive Least Square Algorithm for Initial Alignment of Strapdown Inertial Navigation System

Bidhan Malakar and B.K. Roy

Department of Electrical Engineering National Institute of Technology Silchar, Assam, India bidhan.nits@gmail.com

Abstract. To improve the alignment accuracy and convergence speed of Strapdown inertial navigation system, an initial alignment which is based on Bilinear Recursive Least Square adaptive filter is proposed. The error model for the Strapdown Inertial Navigation System (SINS) is derived from the dynamic model by considering a small misalignment angle. In the literature many algorithms are proposed for the proper estimation of alignment accuracy for INS and it is still a challenge. In this paper, two algorithms which are mainly based on nonlinear adaptive filter viz. Volterra Recursive Least Square (VRLS) and Bilinear Recursive Least Square (BRLS) are proposed and compared for proper estimation of accurate azimuth alignment error. The comparative performances of the aforesaid algorithms are studied and the performance of proposed BRLS algorithm is found to be effective which is obtained in existence of two different white Gaussian noises. The simulation work is done in MATLAB simulating environment. For the realization and validation of proposed BRLS algorithm, the comparative analysis is also precisely presented.

1 Introduction

The Strapdown Inertial Navigation System (SINS) [1] has special advantages and is widely being adopted for the accurate positioning and navigation of missiles, aeroplanes, ships, and railway vehicles etc. INS is advantageous when compared with (Global Positioning System) GPS, as it is unaffected by the external sources. However the INS output is subjective to the errors in the data which is supplied to the system during long durations and also can be due to inaccurate design and construction of the system components. Three types of errors are mainly responsible for the rate at which navigation error grow over long distances of time and these are the initial alignment errors, sensor errors and computational errors. The initial alignment [2-5] is the main key technology in SINS and depending upon the sensors configuration it must provide accurate result. The main purpose of initial alignment is to get the initial coordinate transformation matrix from the body frame to computer coordinates frame and the misalignment angle is considered to be zero during the mathematical modeling. The performance of inertial navigation system is affected by

the alignment accuracy directly as well as initial alignment time which is mainly responsible for the rapid response capability. Therefore there is a requirement of shorter alignment time with a high precision in initial alignment.

At present, Kalman filtering [2] techniques are basically used in order to achieve the initial alignment of inertial navigation system due to its simplicity and also considered as an effective method. But in conventional Kalman filtering technique one must have the future knowledge of the mathematical models [6] and the noise statistics must also be studied and considered. In case of conventional Kalman filter, it is unable to provide a better and efficient result.

After introduction in section 1, section 2 briefly describes initial SINS alignment. The description of estimation algorithms are given in section 3. Section 4 deals with the dynamic modeling used for the simulation. Section 5 deals with the dynamic simulation of fine alignment. Section 6 presents the results, discussions and the comparison between the proposed algorithms. Finally, we conclude the paper in section 7.

2 SINS Initial Alignments

2.1 Coarse Alignment

An INS determines the position of the body frame by the integration of measured acceleration and rotation rate. Since the position is always relative to the starting position and for this reason the INS must know the position, attitude and heading before the navigation begins. The position is assumed to be known but the attitude and heading needs to be determined and is the process of alignment [7], [8].

In the coarse alignment stage the measured acceleration and rotation rate in body frame is compared with the gravity vector G along with the Earth's rotation rate and it directly estimates transformation matrix of carrier coordinates to geographical coordinates. It estimates the attitude and heading accurately enough to justify the small angle approximations made in the error model, and hence allow fine alignment to use an adaptive filter using the error model to obtain a more precise alignment which helps to determine the direction cosine matrix or attitude matrix C_n^b relating the navigation frame (n) and body frame (b). For determining the orientation of the body frame INS makes the use of the accelerometers and gyroscopes for measurement with respect to a reference frame and it is required for the estimation of the measured value of C_n^b .

The basic of alignment in SINS is discussed which plays an important role to improve the initial alignment. Many algorithms are being proposed recently in the literature for estimation and optimization of errors for SINS [9-11]. A theoretical background of the proposed algorithms is discussed in the next section.

3 Theoretical Background

The general set up of an adaptive-filtering [12] environment is illustrated in Fig. 1.



Fig. 1. Adaptive filtering structure

Here k is the iteration number, x(k) denotes the input signal, y(k) is the adaptivefilter output signal, and d(k) defines the desired signal [12]. The error signal e(k) is calculated as d(k)-y(k). In order to determine the proper updating of the filter coefficients, the error signal is then used to form a performance (or objective) function that is required by the adaptation algorithm. The minimization of the objective function implies that the adaptive-filter output signal is matching the desired signal in some sense.

3.1 Bilinear RLS Algorithm (BRLS)

The most widely accepted nonlinear difference equation model used for adaptive filtering is the so-called bilinear equation given by (1).

$$y(k) = \sum_{m=0}^{M} b_m(k) x(k-m) - \sum_{j=1}^{N} a_j(k) y(k-j) + \sum_{i=0}^{I} \sum_{l=1}^{L} C_i, lx(k-i) y(k-l)$$
⁽¹⁾

Here y(k) is the adaptive-filter output and for this case, the signal information vector is defined by (2) and (3).

$$\phi(k) = \left[x(k) \ x(k-1) \dots \ x(k-M) \ y(k-1) \ y(k-2) \ y(k-N) \ x(k)y(k-1) \ x(k-1)y(k-L+1) \ x(k-1)y(k-L)\right]^{T}$$
(2)

$$\theta(k) = \begin{bmatrix} b_0(k) \ b_1(k) \dots \ b_M(k) \ -a_1(k) \ -a_2(k) \dots \ -a_N(k) \ C_{0,1}(k) \ CI, L-1(k) \ CI, L(k) \end{bmatrix}^T$$
(3)

Here N, M, I and L are the orders of the adaptive-filter difference equations.

After the explanation of the two nonlinear adaptive algorithms that are used in the estimation of initial alignment angles error in SINS. The next section presents the dynamic model of SINS.

```
1. Initialization (set parameters for input vector
                     initialization)
    a_i(k) = b_i(k) = c_{i-1}(k) = e(k) = 0.
    y(k) = x(k) = 0, k < 0
    S_d(0) = \delta^{-1}I where \deltacan be the inverse of an estimate of
                     the input signal power
    e(k) is the signal error
    S<sub>d</sub>= inverse of the deterministic correlation matrix
                    of the input vector
    I= identity matrix
2. For each x(k), d(k), k \ge 0, do
     y(k) = \theta^{T}(k)\theta(k)
     e(k) = d(k) - y(k)
     e(k) = signal error, y(k) = adaptive filter output,
     \theta(k) = coefficient vector
3. Calculate e(k) using the relation
    S_D(k+1) = \frac{1}{\lambda} \left[ S_D(k) - \frac{S_D(k)\varphi(k)\varphi^T(k)S_D(k)}{\lambda + \varphi^T(k)S_D(k)\varphi(k)} \right]
    \theta(k+1) = \theta(k) - S_{D}(k+1)\varphi(k)e(k)
```

4 Dynamic Modeling

The model of a local level NED (North-East-Down) [2] frame is used in this paper as the navigation frame. From the alignment point of view the East and North axes are referred to as leveling axes and "Down" axis are called the azimuth axis. The position and velocity errors are ignored. The state equations [2] can be represented as given.

$$\dot{X} = AX + FW = \begin{pmatrix} P_1 & P_2 \\ 0_{5X5} & 0_{5X5} \end{pmatrix} X + \begin{pmatrix} P_2 \\ 0_{5X5} \end{pmatrix} W$$
(4)

where $X = [\delta V_N \ \delta V_E \phi_N \phi_E \phi_D \nabla_x \nabla_y \varepsilon_x \varepsilon_z \varepsilon_z]^T$

 $\delta V_N, \delta V_E$ are the east and north velocity error respectively and ∇_x, ∇_y for accelerometer error. ϕ_N, ϕ_E are the two level milianment angles and ϕ_U for the misalignment angle. $\varepsilon_x, \varepsilon_y, \varepsilon_z$ are the gyro error and ω_w for the Earth rotation rate.

The observation equation for Initial alignment of SINS is given by,

$$Y = TX + V \tag{5}$$

where, $Y = \begin{pmatrix} \delta V_E \\ \delta V_N \end{pmatrix}$, $T = \begin{pmatrix} I_{2X2} \\ 0_{2X8} \end{pmatrix}^T$ and V is the assumed to be the white noise.

The system noise is as follows,

$$W = [w_{ax}w_{ay}w_{gx}w_{gy}w_{gz}]^{T}$$

(

where x, y, z correspond to the Northeast coordinates (NED).

$$P_{1} = \begin{bmatrix} 0 & 2\omega_{ie}Sin(L_{0}) & 0 & -g & 0 \\ -2\omega_{ie}Sin(L_{0}) & 0 & g & 0 & 0 \\ 0 & -\frac{1}{R} & 0 & \omega_{ie}Sin(L_{0}) & -\omega_{ie}Cos(L_{0}) \\ \frac{1}{R} & 0 & -\omega_{ie}Sin(L_{0}) & 0 & 0 \\ \frac{\tan L_{0}}{R} & 0 & \omega_{ie}Cos(L_{0}) & 0 & 0 \end{bmatrix}$$

and,

$$\mathbf{P}_{2} = \begin{pmatrix} \mathbf{C}_{11} & \mathbf{C}_{12} & \mathbf{0} & \mathbf{0} & \mathbf{0} \\ \mathbf{C}_{21} & \mathbf{C}_{22} & \mathbf{0} & \mathbf{0} & \mathbf{0} \\ \mathbf{0} & \mathbf{0} & \mathbf{C}_{11} & \mathbf{C}_{12} & \mathbf{C}_{13} \\ \mathbf{0} & \mathbf{0} & \mathbf{C}_{21} & \mathbf{C}_{22} & \mathbf{C}_{23} \\ \mathbf{0} & \mathbf{0} & \mathbf{C}_{31} & \mathbf{C}_{32} & \mathbf{C}_{33} \end{pmatrix}$$

With the help of transformation matrix i.e. posture matrix P_2 the NED coordinate system can be transformed into body coordinate system and is given by,

 $\mathbf{P}_{2} = \begin{pmatrix} \mathbf{C}_{11} & \mathbf{C}_{12} & \mathbf{C}_{13} \\ \mathbf{C}_{21} & \mathbf{C}_{22} & \mathbf{C}_{23} \\ \mathbf{C}_{31} & \mathbf{C}_{32} & \mathbf{C}_{33} \end{pmatrix}$

$$\begin{split} C_{11} &= \cos(\phi_N)\cos(\phi_U) - \sin(\phi_E)\sin(\phi_N)\sin(\phi_U), \ C_{12} &= \sin(\phi_E)\sin(\phi_N)\cos(\phi_U) + \cos(\phi_N)\sin(\phi_U), \ C_{13} &= -\sin(\phi_N)\cos(\phi_E), \\ C_{21} &= -\cos(\phi_E)\sin(\phi_U), \ C_{22} &= \cos(\phi_E)\cos(\phi_U), \ C_{23} &= \sin(\phi_E), \end{split}$$

 $C_{31} = \sin(\phi_N)\cos(\phi_U) + \cos(\phi_N)\sin(\phi_E)\sin(\phi_U), C_{32} = \sin(\phi_N)\sin(\phi_U) - \cos(\phi_N)\sin(\phi_E)\cos(\phi_U), C_{33} = \cos(\phi_N)\cos(\phi_E)\cos(\phi_E)\cos(\phi_U)$

The systems observation equation by considering level velocity error as an external observation value is

$$Y = \begin{pmatrix} 1 & 0 & 0 & 0 & 0 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 & 0 & 0 & 0 & 0 \end{pmatrix} X + V$$
(6)

The continuous system in equation (4) and (5) must be converted into discrete system before the MATLAB simulation.

After modeling of the SINS the simulation of the model is given in the next section.

5 Dynamic Simulation of Fine Alignment

In order to evaluate the performance of the proposed Volterra RLS and Bilinear RLS algorithm, an example of a static self aligned stationary-base filtering program is

considered, which use the NED coordinate frame [2]. The initial parameters used for the simulation are as follows.

Inertial navigation system location of longitude 125°, north latitude 45°, the initial value X_{θ} of the state variable X are assigned to zero; P₀, Q and R are taken as the corresponding value of the middle-precision gyroscopes and accelerometers; and the initial misalignment $\delta V_N \delta V_E \phi_N \phi_E \phi_U \nabla_X \nabla_y \varepsilon_x \varepsilon_y \varepsilon_z$ angles $\phi_N \phi_E \phi_U$ are taken as 1°; gyro drift are often taken as 0.02°/h, random drift taken as 0.01°/h; and accelerometers taken as 100µg, velocity errors taken as 0.1*m/s*, then

 $X_0 = \text{diag}\{0,0,0,0,0,0,0,0,0,0\}; P_0 = \text{diag}[(0.1 \text{ m/s})^2, (0.1 \text{ m/s})^2, (1^\circ)^2, (1^\circ)^2, (1^\circ)^2, (100\mu g)^2, (100\mu g)^2, (0.02^\circ/h)^2, (0.02^\circ/h)^2, (0.02^\circ/h)^2]; Q = \text{diag}[(50\mu g)^2, (50\mu g)^2, (0.01^\circ/h)^2, (0.01^\circ/h)^2$



Fig. 2. Azimuth MSE plot vs time samples using BRLS and VRLS



Fig. 3. Azimuth error plot vs time samples using BRLS and VRLS



Fig. 4. Azimuth estimation plot vs time samples using BRLS and VRLS



Fig. 5. Azimuth MSE plot vs time samples using BRLS and VRLS



Fig. 6. Azimuth error plot vs time samples using BRLS and VRLS



Fig. 7. Azimuth estimation plot vs time samples using BRLS and VRLS

In this section the various outputs obtained from the MATLAB on a laptop with 1GB RAM and 1.50 GHz processor, are shown and their performances characteristic are discussed in the next section.

6 Results and Discussions

The two algorithms are compared to each other by considering two different values of noises i.e. for $0.05 \ rand(n)$ and $0.03 \ rand(n)$. The performance of two algorithms for estimating Azimuth angle error is shown in Fig. 3, Fig. 6. Azimuth error estimation is shown in Fig. 4 and Fig. 7 and the MSE Azimuth angle error is shown in Fig. 2 and Fig. 5. The comparison performance is given in Table 1.

8

Algorithms	When Noise Covariance is 0.03	When Noise Covariance is 0.05	Computation al Time (Sec)
VRLS	0.630	0.810	0.60624
BRLS	0.372	0.451	0.26374

Table 1. Performance of The Proposed Algorithm for Estimating Azimuth angle Error

7 Conclusions

This paper reports a research on the adaptive filtering technique applied to improve the estimation accuracy of SIN'S azimuth angle error. From the simulation result it is observed that the BRLS filtering algorithm is one of the better and also an effective algorithm for the initial alignment accuracy and convergence speed of strap down inertial navigation system.

References

- 1. Titterton, D., Weston, J. In: Strapdown Inertial Navigation Technology, Institution of Electrical Engineers (2004)
- Wang, X., Shen, G.: A Fast and Accurate Initial Alignment Method for Strapdown Inertial Navigation System on Stationary Base. Journal of Control Theory and Applications, 145– 149 (2005)
- Sun, F., Zhang, H.: Application of a New Adaptive Kalman Filitering Algorithm in Initial Alignment of INS. In: Proceedings of the IEEE International Conference on Information and Automation, Beijing, China, pp. 2312–2316 (2011)
- Gong-Min, V., Wei-Sheng, Y., De-Min, X.: Application of simplified UKF in SINS initial alignment for large misalignment angles. Journal of Chinese Inertial Technology 16(3), 253–264 (2008)
- Anderson, B.D.O., Moore, J.B.: Optimal Filtering. Prentice-Hall Inc., Englewood Cliffs (1979)
- Savage, P.G.: A unified mathematical framework for strapdown algorithm design. J. Guid. Contr. Dyn. 29, 237–249 (2006)
- 7. Silson, P.M.: Coarse alignment of a ship's strapdown inertial attitude reference system using velocity loci. IEEE Transcript Instrumentation Measurement (2011)
- Li, Q., Ben, Y., Zhu, Z., Yang, J.: A Ground Fine Alignment of Strapdown INS under a Vibrating Base. J. Navigation 1, 1–15 (2013)
- Wu, M., Wu, Y., Hu, X., Hu, D.: Optimization-based alignment for inertial navigation systems: Theory and algorithm. Aeros. Science & Technology, 1–17 (2011)
- Salychev, O.S.: Applied Estimation Theory in Geodetic and Navigation Applications. Lecture Notes for ENGO 699.52, Department of Geomatics Engineering, University of Calgary (2000)
- Julier, S.J., Uhlmann, J.K.: Unscented filtering and nonlinear estimation. In: Proc. of the IEEE Aerospace and Electronic Systems, pp. 401–422 (2004)
- 12. Paulo, S.R.: Adaptive Filtering Algorithms and Practical Implementation, 3rd edn. Springer

Time-Domain Solution of Transmission through Multi-modeled Obstacles for UWB Signals

Sanjay Soni¹, Bajrang Bansal¹, and Ram Shringar Rao²

¹Delhi Technological University, Delhi, India {sanjoo.ksoni,bajrangbnsl}@gmail.com ²Ambedkar Institute of Advanced Communication Technologies and Research, Delhi, India rsrao08@yahoo.in

Abstract. In this work, the time-domain solution for transmission through multi-modeled obstacles has been presented. The transmission through dielectric wedge followed by a dielectric slab has been analyzed. The analytical timedomain transmission and reflection coefficients for transmission through the conductor-dielectric interface, considering oblique incidence, are given for both soft and hard polarizations. The exact frequency-domain formulation for transmitted field at the receiver has been simplified under the condition of low-loss assumption and converted to time-domain formulation using inverse Laplace transform. The time-domain results have been validated with the inverse fast Fourier transform (IFFT) of the corresponding exact frequency-domain results. Further the computational efficiency of both the methods is compared.

Keywords: Ultra wideband, Propagation model, Transmission, Frequency-domain, Time-domain.

1 Introduction

In recent years, research in ultra wideband (UWB) propagation through indoor scenario where non line of sight (NLOS) communication is more dominant, has received great attention because of unique properties of UWB communication like resilient to multipath phenomena, good resolution and low power density. In radio propagation of UWB signals, especially in NLOS communication in deep shadow regions, transmitted field component proves to be very significant [1, 2]. Considering the huge bandwidth (3.1-10.6 GHz) of UWB signals, it is more efficient to study UWB propagation directly in time-domain (TD) where all the frequencies are treated simultaneously [3, 4]. The TD solution of transmitted field through a dielectric slab was presented in [5]. A simplified TD model for UWB signals transmitting through a dielectric slab was presented in [6, 7]. The TD solutions for the reflection and transmission through a dielectric slab were presented in [8].

In this paper, we present an approximate TD solution for the field transmitted through multi-modeled obstacles under the low-loss assumption, for UWB signals. In other words, an accurate TD solution for modeling the transmission of UWB signals through multi-modeled obstacles is presented. Transmission model is called multi-
modeled because the transmission is considered through multiple shaped obstacles like a wedge followed by a slab. The analytical TD transmission and reflection coefficients, for transmission through the conductor-dielectric interface, considering oblique incidence, are presented for both soft and hard polarizations. Through these TD formulations, the expression of the TD transmitted field is obtained. The TD results are validated against the IFFT [9, 10] of the corresponding exact frequency-domain (FD) results. At last, the computational efficiency of the two approaches is compared to emphasize the significance of the TD solutions presented.

The paper is organized as follows. In section 2, the propagation environment is presented. Section 3 presents the TD transmission and reflection coefficients and the TD formulations for the computation of the transmitted field component. In section 4, the aforementioned numerical TD results are calculated and validated against the numerical IFFT of the corresponding exact FD results and finally section 5 concludes the paper.

2 Propagation Environment

The propagation environment is shown in Fig. 1, where a single dielectric wedge is followed by a dielectric slab. The parameters r_i , i = 1, 2, ..., 5 are the distances traversed by the transmitted field through the structure from the transmitter (Tx) up to the receiver (Rx).



Fig. 1. Transmission through a dielectric wedge followed by a dielectric slab

Angles $\theta_1, \theta_3, \theta_5, \theta_7$ are the incidence angles with $\theta_2, \theta_4, \theta_6, \theta_8$ as the angles of refraction at points 'P', 'Q', 'R' and 'S' respectively and a_i is the internal wedge angle. The parameters h_t and h_r are the heights of the transmitter and the receiver, h_w is the height of the wedge and h_s is the height of the slab. Tx is at a distance d_1 from the wedge, d_2 is distance between wedge and slab, d_3 is width of slab and d_4 is distance between slab and Rx.

3 Proposed Transmission Model

3.1 TD Transmission Coefficient

The actual refracted angle across conductor-dielectric interface is given by [11]

$$\psi_t(\omega) = \tan^{-1} \left\{ t(\omega) / q(\omega) \right\}$$
(1)

with $t(\omega) = \beta_1(\omega)\sin\theta_1$ and $q(\omega) = s(\omega)\{\alpha_2(\omega)\sin\zeta(\omega) + \beta_2(\omega)\cos\zeta(\omega)\}$.

where α_i and β_i are the attenuation constant and phase shift constant of i^{th} medium. θ_1 is the incident angle and $\cos\{\theta_2(\omega)\} = s(\omega)\exp[j\zeta(\omega)]$ with $\theta_2(\omega)$ as the complex refracted angle.

From (1) it is clear that the true refraction angle is also frequency dependent in nature, which means that for different frequency components of the UWB signal, the true real angles of refraction are different. However for low-loss dielectric obstacles (i.e. $\sigma / \omega \epsilon << 1$), the different true refracted angles reduce to an effective, constant real angle [12]. Thus the real refracted angle ψ_t can be treated as constant and frequency independent, to obtain the approximate analytical TD transmission coefficients. Now for hard polarization case, transmission coefficient while propagating from conductor to dielectric medium, is given by [6, 13, 14]

$$\gamma_h(t) \approx \delta(t) - r_h(t) \tag{2}$$

$$r_{h}(t) = \left[K_{h}\delta(t)\right] + \frac{4k_{h}}{\left(1 - k_{h}^{2}\right)}e^{-pt}\left[\frac{K_{h}}{2}X_{h} + \frac{(1 - X_{h})}{2K_{h}} - \frac{(pt)X_{h}}{4}\right]$$
(3)

with $X_h = e^{-\left(\frac{K_h pt}{2}\right)}, K_h = \left(\frac{1-k_h}{1+k_h}\right), k_h = (\cos\theta_i / \cos\psi_i)(1/\sqrt{\varepsilon_{r2}})$, and $p = \tau/2$ with

 $\tau = \sigma/\varepsilon_2$.

where, ε_2 and εr_2 are dielectric permittivity and relative dielectric permittivity of the dielectric medium respectively. The TD transmission coefficient for a soft polarized wave propagating from conductor to dielectric medium is given as

$$\gamma_{s}(t) \approx \left[\delta(t) - r_{s}(t)\right] \left(\frac{\cos \theta_{i}}{\cos \psi_{t}}\right)$$
(4)

where,

$$r_{s}(t) = K_{s}\delta(t) + \frac{4k_{s}}{(1-k_{s}^{2})}e^{-pt}\left[\frac{K_{s}}{2}X_{s} + \frac{(1-X_{s})}{2K_{s}} - \frac{(pt)X_{s}}{4}\right]$$
 with

$$X_s = e^{-\left(\frac{K_s pt}{2}\right)}, \quad K_s = \left(\frac{1-k_s}{1+k_s}\right), \quad k_s = \left(\cos\psi_t / \cos\theta_i\right) \left(1/\sqrt{\varepsilon_{r2}}\right).$$

3.2 Transmitted Field through the Propagation Environment

The FD expression for the transmitted field at Rx through the structure shown in Fig. 1 is given by [2, 12]

$$E_{RX}(\omega) = \left(E_i(\omega) / r_{total}(\omega)\right) \left(\prod_{i=1}^4 T_{i,s,h}(\omega)\right) \left(\exp(-jk_0r_1)\prod_{j=2}^5 \exp\left\{-\left(\alpha_{ej}(\omega) + j\beta_{ej}(\omega)\right)\right\}\right)$$
(5)

with $r_{total}(\omega) = \sum_{i=1}^{5} r_i$ and $T_{total,s,h}(\omega) = \prod_{i=1}^{4} T_{i,s,h}(\omega)$ where $T_{i,s,h}(\omega)$, i = 1, 2, ..., 4 are

the FD transmission coefficients with respect to points 'P', 'Q', 'R' and 'S' (See Fig. 1). $\alpha_{ej}(\omega), \beta_{ej}(\omega)$ are total effective attenuation constants and phase shift constants for different j^{th} regions [12]. Actual FD path-loss expression from (5) is given by

$$L_{total,s,h}(\omega) = \left(\exp(-jk_0r_1)\prod_{j=2}^5 \exp\left\{-\left(\alpha_{ej}(\omega) + j\beta_{ej}(\omega)\right)\right\}\right)$$
(6)

Now the corresponding TD expression for the received field at Rx based on the FD transmission model of [2] is as follows

$$e_{RX}(t) \approx \left(\frac{e_i(t)}{r_{total}}\right) * \Gamma_{1,s,h}(t) * \Gamma_{2,s,h}(t) * \Gamma_{3,s,h}(t) * \Gamma_{4,s,h}(t) * l_{total,s,h}(t)$$
(7)

with '*' representing the convolution operator, $\Gamma_{1,s,h}(t) * \Gamma_{2,s,h}(t) * \Gamma_{3,s,h}(t) * \Gamma_{4,s,h}(t) = \Gamma_{total,s,h}(t)$ is the TD counterpart of $T_{total,s,h}(\omega)$, $l_{total,s,h}(t)$ [6, 7] is the TD counterpart of $L_{total,s,h}(\omega)$. The TD expressions for $\Gamma_{i,s,h}$, i = 1, 2, ..., 4 can be obtained using (2) and (4) for different polarizations.

For loss tangent much less than unity ($\sigma / \omega \epsilon \ll 1$), the FD path-loss expression (6) reduces to the following approximate form with constant values of angles of refraction and along a single effective path for transmission:

$$L_{total,s,h}(\omega) \approx \exp\left(-jk_0(r_1 + r_3 + r_5)\right) \exp\left[-j\omega\sqrt{\mu\varepsilon}\left(1 + \frac{\sigma}{2j\omega\varepsilon}\right)\left\{r_2 + \left(d3\sqrt{\frac{\varepsilon_r}{\varepsilon_r - \sin^2\theta_5}}\right)\right\}\right]$$
(8)

Here $r_1 + r_3 + r_5$ is the total distance travelled by the field in free space. ε and σ are the parameters of the dielectric mediums in Fig. 1 (assuming same for wedge and slab). The term $l_{total,s,h}(t)$ in (7) is then given by

$$l_{total,s,h}(t) \approx \exp\left[-\sqrt{\frac{\mu}{\varepsilon}} \left(\frac{\sigma}{2}\right) \left\{ r_2 + \left(d_3 \sqrt{\frac{\varepsilon_r}{\varepsilon_r - \sin^2 \theta_5}} \right) \right\} \right]$$

$$\delta\left[t - \sqrt{\mu\varepsilon} \left\{ r_2 + \left(d_3 \sqrt{\frac{\varepsilon_r}{\varepsilon_r - \sin^2 \theta_5}} \right) \right\} \right] * \delta\left(t - \frac{r_1 + r_3 + r_5}{c} \right)$$
(9)

This approximated TD path-loss expression will be used in (7) to compute the TD transmitted field and the accuracy will be proved by the comparison of TD transmitted field with the IFFT of the exact FD results, as shown in next section.

4 **Results and Discussions**

In this section, our goal is to compare the proposed TD solution with conventional IFFT-FD method. Table 1 shows the electromagnetic properties of the considered materials.

 Table 1. Electromagnetic properties of different dielectric materials

Material	Relative Permittivity	Conductivity (S/ m)
Wood[6]	2	0.01
Drywall[15]	2.4	0.004
Glass[15, 16]	6.7	0.001

Fig. 2 shows the transmitted field through the propagation environment discussed in section 2, for both hard and soft polarizations. The transmitted field at Rx suffers no distortion in shape in comparison to the shape of the excited UWB pulse. This is because of small magnitude of the loss tangent with respect to unity. However, the amplitude of the transmitted field is attenuated because of the transmission loss through the dielectric mediums.



Fig. 2. Transmitted field through 'dielectric wedge followed by a dielectric slab', with glass [15, 16]

The TD results for the transmitted field for both the polarizations are in excellent agreement with corresponding IFFT of exact FD results, thus providing validation to the proposed TD solution.

Fig. 3 shows the effect of varying Rx position (changing distance d_4 in Fig. 1) on transmitted field at the receiver. Transmitted field gets more attenuated as Rx moves away from the obstacles. The results for soft and hard polarized fields come closer to each other as the distance d_4 increases. Also the TD results match closely with the IFFT-FD results.



Fig. 3. Transmitted field through 'dielectric wedge followed by a dielectric slab' for different receiver positions, with glass [15, 16]



Fig. 4. Transmitted field through 'dielectric wedge followed by a dielectric slab' for different dielectric materials, with wood [6], drywall [15] and glass [15, 16]

Fig. 4 shows transmitted field at the receiver for different dielectric materials. The TD results are in good agreement with the IFFT-FD results. It can be seen that as the value of loss tangent decreases, a better agreement is achieved between the TD and IFFT-FD results.

A comparison between the computation times of the IFFT-FD method and the proposed TD solution for propagation profile considered in Fig. 1 is presented in Table 2. The presented results in Table 2 establish that the proposed TD analysis is computationally very efficient in comparison to the IFFT-FD solution.

Propagation profile	$T_{IFFT-FD}/T_{TD}$
For soft polarization	~198
For hard polarization	~191

Table 2. Efficiency comparison of two methods

The two main reasons for such a significant reduction in the computational time in TD are: (i) the efficient convolution technique [9] due to which few number of time samples suffice to provide accurate results. (ii) Approximation of the multiple transmission paths in FD by a single effective path for low-loss dielectric case.

Given the excellent agreement between proposed TD solution and IFFT of FD solution, it can be concluded that the proposed method is accurate for low loss tangent values in the UWB bandwidth. The presented work also establishes that the proposed TD solution is computationally more efficient than the conventional IFFT-FD method.

5 Conclusion

An analytical TD solution has been presented for the transmitted field through multi-modeled obstacles made up of low-loss dielectric materials. Analytical TD transmission and reflection coefficients for transmission through an interface between conductor and dielectric mediums are presented for soft and hard polarizations. The results of the proposed TD solution are validated against the corresponding IFFT-FD results and the computational efficiency of two methods is compared. The TD solution outperforms the IFFT-FD analysis in terms of the computational efficiency. The TD solution is vital in the analysis of UWB communication as it can provide a fast and accurate prediction of the total transmitted field in microcellular and indoor propagation scenarios.

References

- de Jong, Y.L.C., Koelen, M.H.J.L., Herben, M.H.A.J.: A building-transmission model for improved propagation prediction in urban microcells. IEEE Trans. Veh. Technol. 53(2), 490–502 (2004)
- Soni, S., Bhattacharya, A.: An analytical characterization of transmission through a building for deterministic propagation modeling. Microw. Opt. Techn. Lett. 53(8), 1875–1879 (2011)
- Karousos, A., Tzaras, C.: Multiple time-domain diffraction for UWB signals. IEEE Trans. Antennas Propag. 56(5), 1420–1427 (2008)

- 4. Qiu, R.C., Zhou, C., Liu, Q.: Physics-based pulse distortion for ultra-wideband signals. IEEE Trans. Veh. Technol. 54(5), 1546–1555 (2005)
- Chen, Z., Yao, R., Guo, Z.: The characteristics of UWB signal transmitting through a lossy dielectric slab. In: Proc. IEEE 60th Veh. Technol. Conf., VTC 2004-Fall, Los Angeles, CA, USA., vol. 1, pp. 134–138 (2004)
- Yang, W., Qinyu, Z., Naitong, Z., Peipei, C.: Transmission characteristics of ultra-wide band impulse signals. In: Proc. IEEE Int. Conf. Wireless Communications, Networking and Mobile Computing, Shanghai, pp. 550–553 (2007)
- Yang, W., Naitong, Z., Qinyu, Z., Zhongzhao, Z.: Simplified calculation of UWB signal transmitting through a finitely conducting slab. J. Syst. Eng. Electron. 19(6), 1070–1075 (2008)
- Karousos, A., Koutitas, G., Tzaras, C.: Transmission and reflection coefficients in timedomain for a dielectric slab for UWB signals. In: Proc. IEEE Veh. Technol. Conf., Singapore, pp. 455–458 (2008)
- 9. Brigham, E.O.: The Fast Fourier transform and Its Applications. Prentice-Hall, Englewood Cliffs (1988)
- Sevgi, L.: Numerical Fourier transforms: DFT and FFT. IEEE Antennas Propag. Mag. 49(3), 238–243 (2007)
- 11. Balanis, C.A.: Advanced engineering electromagnetic. Wiley, New York (1989)
- Tewari, P., Soni, S.: Time-domain solution for transmitted field through low-loss dielectric obstacles in a microcellular and indoor scenario for UWB signals. IEEE Trans. Veh. Technol. (2013) (under review)
- Barnes, P.R., Tesche, F.M.: On the direct calculation of a transient plane wave reflected from a finitely conducting half space. IEEE Trans. Electromagn. Compat. 33(2), 90–96 (1991)
- Tewari, P., Soni, S.: A comparison between transmitted and diffracted field in a microcellular scenario for UWB signals. In: Proc. IEEE Asia-Pacific Conf. Antennas Propag., Singapore, pp. 221–222 (2012)
- Muqaibel, A., Safaai-Jazi, A., Bayram, A., Attiya, A.M., Riad, S.M.: Ultrawideband through-the-wall propagation. IEE Proc.-Microw. Antennas Propag. 152(6), 581–588 (2005)
- Jing, M., Qin-Yu, Z., Nai-Tong, Z.: Impact of IR-UWB waveform distortion on NLOS localization system. In: ICUWB 2009, pp. 123–128 (2009)

Indexing and Retrieval of Speech Documents

Piyush Kumar P. Singh, K.E. Manjunath, R. Ravi Kiran, Jainath Yadav, and K. Sreenivasa Rao

Indian Institute of Technology Kharagpur, West Bengal - 721302, India piyushks@live.com, {ke.manjunath,r.ravi.kiran.88,jaibhu38}@gmail.com, ksrao@iitkgp.ac.in

Abstract. In this paper, a speech document indexing system and similarity-based document retrieval method has been proposed. *K*-d tree is used as the index structure and codebooks derived from speech documents present in the database, are used during retrieval of desired document. Each document is represented as a sequence of codebook indices. The longest common subsequence based approach is proposed for retrieving the documents. Proposed retrieval method is evaluated using a speech database of 3 hours recorded by a male speaker and speech queries from 5 male and 5 female speakers. The accuracy of retrieval is found to be about 88% for the queries given by male speakers.

Keywords: Indexing and Retrieval, codebook, MFCC, k-d tree, retrieval, longest common subsequence.

1 Introduction

Due to rapid advancement in technology, there has been explosive growth in the generation and use of multimedia data, such as video, audio, and images. A lot of audio and video data is generated by internet, mobile devices, TV and radio broadcast channels. Indexing systems are desirable for managing and supporting usage of large databases of multimedia. Audio indexing finds applications in digital libraries, entertainment industry, forensic laboratories and virtual reality.

Many kinds of indexing schemes have been developed, and studied by researchers working in this field [13]. An approach for indexing audio data is the use of text itself. Transcripts from the audio data are generated, which are used for indexing. This approach is effective for indexing broadcast news, video lectures, spoken documents, etc., where the clean speech data is available. The retrieval from such indexing system is performed using keywords as a query. For effective use of multimedia data, the users should be able to make content based queries or queries-by-example, which are unrestricted and unanticipated. Content based retrieval systems accept data type queries i.e., hummed, sung or original clip of a song for a song retrieval. The methods used for audio indexing can be broadly classified as under:

 Signal parameter based systems [1,3,9,14] - In this scheme, the signal statistics such as mean, variance, zero crossing rate, autocorrelation, histograms of samples/difference of samples, energy contour, loudness contour, etc. either on the whole audio signal data or blocks of data are used. This type of indexing supports query by example. Audio Fingerprinting [1] is an example of this type of system.

- Musical parameter based systems [2,7,8,10] - In this scheme, the parameters of signal along with acoustical attributes such as melody contour, rhythm, tempo, etc. extracted from the audio signal are used for indexing. This type of indexing scheme supports both query by example and query by humming. This method requires extensive calculations.

In the design of an audio indexing system following two aspects need to be addressed. (1). Derivation of good features from the audio data to be used as indices during search. (2). Organization of these indices in a suitable multidimensional data structure with efficient search. Selection of a good measure of similarity (distance measure). The objective of this work is to design a contentbased speech indexing system which supports query by example. In this work, the features used for indexing purpose is a codebook derived from MFCC feature vectors for every 10 seconds of speech data present in the database. The k-d tree is used for providing the indexing structure [4] and longest common subsequence (LCS) [11], [5] has been used as a measurement for similarity for retrieval purpose.

Rest of the paper is organized as follows - Section II describes the framework of the proposed indexing and retrieval system. In Section III, experimental results and their analysis are discussed. The summary of the contents presented in the paper, and conclusions drawn from the observations are presented in Section IV.

2 Proposed Speech Indexing and Retrieval System

A speech file is a concatenated sequence of sounds. So finding a matching speech clip in the database to a query clip provided, can be thought of as finding the clip having the similar sounds in the same sequence. But since the query can be provided by any person, the two sequences will not be the exact duplicate of each other. This problem of matching the sequence of sounds can be mapped to the well known problem of determining the longest common subsequence from two sequences of characters. The only difference is that in place of characters there are sounds. By mapping the sound sequence matching problem to the longest common subsequence problem, a different issue arises. How do we map sounds to characters? Determining the longest common subsequence directly from the sounds of the speech files is very difficult because a speaker can produce many variants of a single sound. These variations increase tremendously with change in speaker. So we build codebooks to bring uniformity in these sound sequences. The process to derive a codebook is explained in Section 2.1.

The framework of the proposed speech indexing and retrieval system is shown in the Fig. 1. The tasks of the system and the challenges they pose, can be explained in three phases.



Fig. 1. Proposed speech indexing and retrieval system

Building the Codebook. Phase 1 (Preprocessing phase) converts all raw speech data into indexable items, typically, points in a high-dimensional vector space with an appropriate distance measure. In other words, in this phase we extract features from the speech files and the query to be processed. The features considered here are Mel-Frequency Cepstral Coefficients (MFCCs). MFCCs are determined from speech signal using the following steps:

- 1. Pre-emphasize the speech signal.
- 2. Divide the speech signal into sequence of frames with a frame size of 20 ms and a shift of 10 ms. Apply the Hamming window over each of the frames.
- 3. Compute magnitude spectrum for each windowed frame by applying DFT.
- 4. Mel spectrum is computed by passing the DFT signal through mel filter bank.
- 5. DCT is applied to the log mel frequency coefficients (log mel spectrum) to derive the desired MFCCs.

From each frame, 13 MFFCCs are extracted using 24 filter banks. This results in 1000 MFCC vectors for 10 seconds of speech. These 1000 vectors are then converted to a vector codebook using k-means clustering [12].

Fig. 2 shows the basic steps for building a vector quantization codebook and extraction of code book indices. Training data consists of 1000 MFCC vectors. The training data set is used to create an optimal set of codebook vectors for representing the spectral variability observed in the training data set. A centroid computation procedure is used as the basis of partitioning the training data set



Fig. 2. Codebook building process and codebook indices extraction

into N clusters. Thus it forms an N-center codebook. Here, K-means clustering method is used for clustering the training data. Finally, nearest-neighbor labeling has been used to obtain a sequence of codebook indices. The size of codebook varies with the number of centers used for k-means clustering. The results shown in this paper have been generated on the codebook having 32 centers. A separate codebook is created for every 10 seconds of speech data in database.

Building the Index. Phase 2 (Indexing / Lookup phase) does the task of indexing the features extracted from the speech database. This phase also performs a lookup in the index to retrieve the best matching items for a query vector. The matching can be done in various ways, say, exact matching, nearest neighbour matching, etc. which depends on the application and the efficiency required.

In this work, MFCC feature vectors derived from speech consists of 13 dimensions. Various data structures can be used for multi-dimensional indexing, for example, quad tree, k-d tree, optimized k-d tree. In this work, the index is created from the codebook in the form of a k-d tree. The discriminating key for each level of the k-d tree is selected according to Bentleys definition of the k-d tree i.e., $D = L \mod k + 1$, where D is the discriminating key number for level L and the root node is defined to be at level zero. However, the selection of the partition value is done in the way suggested by Friedman [4]. The partition value is selected as the median value in the dimension of the discriminator(D).

The codebooks created in this work have 32 codebook vectors. Each of these vectors is assigned an index number (1-32). This numbering of the codebook entries is utilized in Phase 3. We are required to preserve this numbering even when the codebook is converted to a k-d tree index. This is achieved by appending the index number as an additional dimension to the codebook vectors while creating the k-d tree. This 14th dimension is never used as a discriminating key while building the k-d tree index. The Algorithm 1 is used for k-d tree creation in this work.

The procedure median(j, subfile) returns median of the j^{th} key values. The procedures $make_terminal$ and $make_non_terminal$ store their parameters as values of the node in the k-d tree and return a pointer to that node. The *leftsubfile* and *rightsubfile* procedures, partition the file along the d^{th} key with respect

A	Algorithm 1. Algorithm used for k-d tree creation		
Result : $root \leftarrow build_tree(entirefile, 1)$			
	magazing huild track file dime)		

```
1 procedure build_tree(file,dim)
 2
        local p
 3
        if size(file) < b then
 \mathbf{4}
             return (make_terminal(file))
 5
        end
        p \leftarrow median(dim, file)
 6
        dim \leftarrow dim \operatorname{Mod} k + 1
 7
        left \leftarrow build\_tree(left\_subfile(d, p, file))
 8
        right \leftarrow build\_tree(right\_subfile(d, p, file))
 9
        return make_non_terminal(d, p, left, right)
10
11 end procedure
```

to the partition value p and return left and right subfiles, respectively. Heapsort algorithm having complexity of O(NlogN), is used to compute the median as well as the left and right sub file at each level. Thus the time complexity of building a k-d tree from N vectors is $O(Nlog^2N)$. The searching algorithm proposed by Friedman [4] is used in this work for searching the k-d tree. This search algorithm is also a part of Phase 2 of the system.

Evaluation of Matches. Phase 3 (Retrieval phase) evaluates all the matching items and decides which files among the database are the best candidates, to be returned as the query response. As mentioned earlier, in this work we use longest common subsequence determination technique for finding the matching speech fragments.

The sequence of codebook indices is generated by using the MFCC feature vector sequence of the speech clip. For each MFCC vector, the index number (1-32) of nearest codebook vector is determined by searching the k-d tree. These



Fig. 3. Retrieval using longest common subsequence approach

index numbers are then concatenated to form a sequence of index numbers. The retrieval method finds out that clip in the database, whose MFCC vectors follow the similar codebook vector sequence, as that of the query clip. Steps in retrieval method are as follows:

- 1. After building the codebook for a speech document in the database, the MFCC vectors are scanned, in order of their occurrence, to determine the index number (1-32) of the nearest codebook vector.
- 2. By concatenating these index numbers obtained in the above step, we obtain an approximate sequence of sounds present in the clip.
- 3. MFCC vectors are extracted from the query speech clip provided.
- 4. By using the same technique of nearest neighbour search among the codebook vectors, we can obtain another sequence of index numbers. The length of the longest common subsequence between the sequences obtained in steps 2 and 4 is determined.
- 5. The length of the longest common subsequence obtained in step 4, can act as the similarity measure while comparison with other clips in the database, since the process of sequence determination will be repeated for all the clips in the database.

Longest Common subsequence is a classic problem in computer science. Many solutions are available in literature. In this work, longest common subsequence is determined by using dynamic programming technique (DP). DP is well known for its space efficient implementation and lower complexity. The entire process is summarized in Fig. 3.

3 Performance Evaluation

The database used for evaluation, consists of 3 hours of speech recorded by a single male speaker. The speech consists of articles about History of India available on Wikipedia. This recorded speech data was divided into 1080 speech documents, each of duration 10 seconds. Proposed retrieval approach was evaluated by 200 queries recorded from 5 male and 5 female speakers. In this work, we have asked each speaker to speak 20 speech documents from the speech database at random. The retrieval performance of the proposed method is shown in Table 1. From the results presented in Table 1, it is observed that the accuracy of retrieval is 88% for the male speaker queries and 3% for female speaker queries. The reason behind the queries being missed is that the speakers are different, and thus their utterances of the similar text may differ. These differences account for the differences in the sequences we are matching. Therefore, if the utterance differs at many places, the sequences get altered immensely resulting in query misses. The gender effect on the MFCC vectors, also plays a role in poor retrieval performance of the female speaker queries.

Speaker Gender						
Male		Female				
Speaker Id	Queries matched	Speaker Id	Queries matched			
1	18	6	0			
2	20	7	0			
3	15	8	2			
4	20	9	0			
5	15	10	1			

Table 1. Retrieval performance with longest common subsequence matching

The poor retrieval performance for the queries of female speakers may be due to the influence of gender characteristics. As the database contains only male speaker's speech utterances, therefore in case of female speakers queries, the indices generated by the queries is entirely different due to differences in shape and size of the vocal tract between the genders.

4 Summary and Conclusions

In this paper, a k-d tree based speech document indexing system has been proposed. For retrieving the desired speech document for a given query, the sequence of codebook indices, generated by the speech document and the query are compared using LCS approach. By computing the LCS scores between a query and all the speech documents, the retrieval system retrieves the desired document based on the highest LCS score. For evaluating the proposed retrieval approach, 3 hours of speech database recorded by a male speaker was used. The performance accuracy in the retrieval process is found to be better for the queries spoken by male speakers. In the case of female speakers, the performance is observed to be very poor.

In this work, the codebooks are generated from each 10 second speech document. The codebook captures the local characteristics of speech present in that 10 second segment. Therefore when we generate the indices from this codebook, the sequence of indices may be differ for different speakers even if the spoken message content is same. This problem may be addressed by building a single codebook from large amount of speech data instead of smaller codebooks for each 10 second segment. From the generalized codebook, if we derive the sequence of indices for the speech document and the query, the retrieval accuracy may be improved. It may also resolve the problem risen due to gender dependent query. Query gender dependency may also be resolved by adapting appropriate vocal tract length normalization(VTLN) feature transformation techniques in the codebook building procedure. Also, in the current experimental setup the queries given were similar to clips present in the speech database. In future, work can be done to perform retrieval based on queries having only few keywords and other connecting words.

References

- Cha, G.-H.: An Effective and Efficient Indexing Scheme for Audio Fingerprinting. In: Proceedings of the 2011 Fifth FTRA International Conference on Multimedia and Ubiquitous Engineering, Washington, DC, USA, pp. 48–52 (2011)
- Chen, A.L.P., Chang, M., Chen, J., Hsu, J.-L., Hsu, C.-H., Hua, S.Y.S.: Query by music segments: an efficient approach for song retrieval. In: 2000 IEEE International Conference on Multimedia and Expo (2000)
- Foote, J.T.: Content-Based Retrieval of Music and Audio. In: Proceedings of SPIE, Multimedia Storage and Archiving Systems II, pp. 138–147 (1997)
- Friedman, J.H., Bentley, J.L., Finkel, R.A.: An Algorithm for Finding Best Matches in Logarithmic Expected Time. ACM Transactions on Mathematical Software 3(3), 209–226 (1977)
- 5. Hirschberg, D.S.: A linear space algorithm for computing maximal common subsequences. ACM Communications, 341–343 (1975)
- Deller Jr., J.R., Hansen, J.H.L., Proakis, J.G.: Discrete-Time Processing of Speech Signal. IEEE Press (2000)
- Kosugi, N., Nishihara, Y., Sakata, T., Yamamuro, M., Kushima, K.: A practical query-by-humming system for a large music database. In: Proceedings of the Eighth ACM International Conference on Multimedia, pp. 333–342 (2000)
- 8. Lemström, K., Laine, P.: Musical information retrieval using musical parameters. In: Proceedings of the 1998 International Computer Music Conference (1998)
- Li, G., Khokhar, A.A.: Content-based indexing and retrieval of audio data using wavelets. In: 2000 IEEE International Conference on Multimedia & Expo, pp. 885– 888 (2000)
- Lu, L., You, H., Zhang, H.-J.: A new approach to query by humming in music retrieval. In: ICME 2001, pp. 595–598 (2001)
- Maier, D.: The Complexity of Some Problems on Subsequences and Supersequences. J. ACM, 322–336 (1978)
- Rabiner, L., Juang, B.-H.: Fundamentals of speech recognition. Prentice-Hall, Inc. (1993)
- Rao, K.S., Pachpande, K., Vempada, R.R., Maity, S.: Segmentation of TV broadcast news using speaker specific information. In: NCC 2012, pp. 1–5 (2012)
- Subramanya, S.R., Youssef, A.: Wavelet-based Indexing of Audio Data in Audio/Multimedia Databases. In: Proceedings of MultiMedia Database Management Systems (1998)

An Improved Filtered-x Least Mean Square Algorithm for Acoustic Noise Suppression

Asutosh Kar¹, Ambika Prasad Chanda¹, Sarthak Mohapatra¹, and Mahesh Chandra²

¹Department. of Electronics and Telecommunication Engineering Indian Institute of Information Technology, Bhubaneswar, India ²Department of Electronics and Communication Engineering Birla Institute of Technology, Mesra, India {asutosh, b210038}@iiit-bh.ac.in, chanda.iiit@gmail.com, shrotriya@bitmesra.ac.in

Abstract. In the modern age scenario noise reduction is a major issue, as noise is responsible for creating disturbances in day-to-day communication. In order to cancel the noise present in the original signal numerous methods have been proposed over the period of time. To name a few of these methods we have noise barriers and noise absorbers. Noise can also be suppressed by continuous adaptation of the weights of the adaptive filter. The change of weight vector in adaptive filters is done with the help of various adaptive algorithms. Few of the basic noise reduction algorithms include Least Mean Square algorithm, Recursive Least Square algorithm etc. Further we work to modify these basic algorithms so as to obtain Normalized Least Mean Square algorithm, Fractional Least Mean Square algorithm, Differential Normalized Least Mean Square algorithm, Filtered-x Least Mean Square algorithm etc. In this paper we work to provide an improved approach for acoustic noise cancellation in Active Noise Control environment using Filtered-x LMS (FXLMS) algorithm. A detailed analysis of the algorithm has been carried out. Further the FXLMS algorithm has been also implemented for noise cancellation purpose and the results of the entire process are produced to make a comparison.

Keywords: adaptive filter, active noise control, Least Mean Square, Mean Square Error, FXLMS.

1 Introduction

Active noise control (ANC) is based on the principle of destructive superposition of acoustic waves [1, 2]. Using ANC concept in noise cancellation, a noise signal is generated which is correlated to the noise signal but opposite in phase. By adding both the original signal can be made noise free. The generation of the signal is controlled by an adaptive algorithm which adaptively changes the weight of the filter used. Active noise control is mainly classified into two categories i.e. "feed-forward ANC" and "feedback ANC". In feedback ANC method a controller is used to modify the response of the system. But in feed-forward ANC method a controller is used to

adaptively calculate the signal that cancels the noise. In this paper "feed-forward ANC" approach is implemented to cancel the noise with the help of FxLMS algorithm. This is because the amount of noise reduction achieved by feed-forward ANC system is more than that of feedback ANC system. A single channel feedforward ANC system comprises of a reference sensor, a control system, a cancelling loudspeaker and an error sensor. The primary noise present is measured by the reference sensor and is cancelled out around the location of the error microphone by generating and combining an anti-phase cancelling noise that is correlated to the spectral content of the unwanted noise [3]. The reference sensor produces a reference signal that is "feed-forward" to the control system to generate "control signals" in order to drive the cancelling loudspeaker to generate the cancelling noise. The error microphone measures the residual noise after the unwanted noise and controls signal. It combines and sends an error signal back to the controller to adapt the control system in an attempt to further reduction of error. The control system adaptively modifies the control signal to minimize the residual error. The most famous adaptation algorithm for ANC systems is the FXLMS algorithm [4, 5] as mentioned earlier, which is a modified and improved version of the LMS algorithm [6-10].

Although FXLMS algorithm is widely used due to its stability but it has slower convergence. Various techniques have been proposed to improve the convergence of FXLMS algorithm with some parameter trade-offs. Other versions of the FXLMS include Filtered-x Normalized LMS (FXNLMS) [11], Leaky FXLMS [12], Modified FXLMS (MFXLMS) [13-15] etc. However, the common problem with all of these algorithms is the slow convergence rate, especially when there is large number of weights [16]. To overcome this problem, more complex algorithms such as Filtered-x Recursive Least Square (FXRLS) [17] can be used. These algorithms have faster convergence rate compared to the FXLMS; however they involve matrix computations and their real-time realizations might not be cost effective.

2 **Problem Formulation**

Acoustic noise problems are not only based on high frequency noise but many are also dominated by low frequency noise. Various passive noise control techniques such as noise barriers and absorbers do not work efficiently in low frequency environments. ANC works very efficiently in case of low frequency noise & it is also cost effective than bulky, heavy barriers and absorbers.

The basic LMS algorithm fails to perform well in the ANC framework. This is due to the assumption made that the output of the filter is the signal perceived at the error microphone, which is not the case in practice. The presence of the A/D, D/A converters, actuators and anti aliasing filter in the path from the output of the filter to the signal received at the error microphone cause significant change in the output signal. This demands the need to incorporate the effects of this secondary path function in the algorithm. But the convergence of the LMS algorithm depends on the phase response of the secondary path, exhibiting ever-increasing oscillatory behavior as the phase increases and finally going unstable at 90° [11]. The solution to this

problem was either to employ an inverse filter in the cancellation path or to introduce a filter in the reference path, which is ideally equal to the secondary path impulse response. The former technique is referred to as the "filtered-error" approach, while the later is now known as the "filtered-reference" method, more popularly known as the FXLMS algorithm. The FXLMS solution is by far the most widely used due to its stable as well as predictable operation [3-5].

3 Simulation Setup

1) Feed-forward ANC system:

The basic idea of feed-forward ANC is to generate a signal (secondary noise), that is equal to a disturbance signal (primary noise) in amplitude and frequency, but has opposite phase. Combination of these signals results in cancellation of the primary (unwanted) noise. This ANC technique is well-known for its use in cancelling unwanted sound as shown in [2].

2) ANC system description using adaptive filter:

In this paper instead of the adaptive algorithm block an improved FXLMS algorithm approach is employed to suppress noise. The most popular feed-forward control algorithm for acoustic noise cancellation is FXLMS algorithm [7] which is the main area of focus in this paper. A control signal is created by the FXLMS algorithm by filtering the reference signal x(n) with an adaptive control filter. The control filter is updated via a gradient descent search process until an ideal filter that minimizes the residual noise found. This is because the existence of a filter in the auxiliary and the error- path is shown which generally degrade the performance of the LMS algorithm. Thus, the convergence rate is lowered, the residual power is increased and the algorithm can even become unstable. In order to stabilize LMS algorithm, the reference signal x(n) is filtered by an estimate of the secondary path transfer function S(n) which is the propagation path from the controller to the error sensor giving the filtered-x signal. Hence the algorithm is known as FXLMS algorithm.

The noise measured at the reference sensor propagates through the primary path represented by P(n) and arrives at the error sensor or receiver as the signal, d(n). This is the unwanted noise to be cancelled which is sometimes referred to as the "desired" signal, meaning the signal that the controller is trying to duplicate with opposite phase. The desired signal will be correlated with the noise characterized by the reference. The LMS algorithm then generates control signal at the cancelling loudspeaker by filtering the reference signal x(n) with an adaptive FIR control filter, W(n). The control signal, y(n), is the convolution of x(n) with W(n). The control signal is filtered by the secondary path transfer function, S(n), and arrives at the error sensor as the output signal, y'(n). The output signal combines with the unwanted noise to give the residual error signal, e(n), measured by the error sensor. An adaptive process searches for the optimal coefficients for the control filter which minimizes the error. This optimal filter is designated W_{ant} [7-10].

Step:1

For ANC system containing a Secondary Path transfer function S(n) the residual error can be expressed as:

$$e(n) = d(n) - y'(n) \tag{1}$$

where, y'(n) is the output of the secondary path S(n).

Step:2

If S(n) be an IIR filter with denominator coefficients $[a_0, \dots, a_n]$ and numerator coefficient $[b_0, \dots, b_{M-1}]$, then the filter output y'(n) can be written as the sum of the filter input y(n) and the past filter output:

$$y'(n) = \sum_{i=1}^{N} a_i y'(n-i) + \sum_{j=0}^{M-1} b_j y(n-j)$$
(2)

Step:3

Hence the gradient estimate becomes:

$$\nabla \dot{\boldsymbol{\xi}}(n) = -2x'(n)e(n), \tag{3}$$

where,

$$x'(n) = \sum_{i=1}^{N} a_i x'(n-i) + \sum_{j=0}^{M-1} b_j x(n-j)$$
(4)

In practical applications, S(n) is not exactly known, therefore the parameters a_i and

 b_i are the parameters of the Secondary Path Estimate S(n).

Step:4

The weight update equation of the FXLMS algorithm is:

$$w(n+1) = w(n) + \mu x'(n)e(n)$$
 (5)

Equation (5) is the filter weight update equation of the FXLMS algorithm. As it can be seen that the equation is identical to the LMS algorithm but here instead of using x(n) i.e. the reference signal, x'(n) is used which is the filtered output of the original signal called as "filtered-x signal" or "filtered reference signal".

The FXLMS algorithm is very tolerant to modelling errors in the secondary path estimate. The algorithm will converge when the phase error between S(n) and H(n) is smaller than +/- 90 degrees. Computer simulations show that phase errors less than 45° do not significantly affect performance of the algorithm [8]. The gain applied to the reference signal by filtering it with S(n) does not affect the stability of the algorithm and is usually compensated for by modifying the convergence parameter, μ . Convergence will be slowed down though, when the phase error increases. H(n) is estimated through a process called system identification. Band-limited white noise is played through the control speaker(s) and the output is measured at the error sensor.

The measured impulse response is obtained as a FIR filter S(n) in the time domain. The coefficients of S(n) are stored and used to pre-filter the reference signal and give the input signal to the LMS update.

The performance of the FXLMS adaptation process is dependent on a number of factors. Those factors include the characteristics of the physical plant to be controlled, the secondary path impulse response and the acoustic noise band-width. Hence the FXLMS will not function properly if the secondary path has a long impulse response and/or the acoustic noise has a wide band-width. Among the design parameters, steady state performance increases, increasing the filter length but convergence of FXLMS algorithm is degraded by this. So the filter length should be chosen carefully. After the filter length is set, the maximum achievable performance is limited by a scalar parameter called the adaptation step-size. The speed of convergence will generally be higher when choosing a higher sample frequency and higher value of μ . [14-16]

It is advantageous to choose a large value of μ because the convergence speed will increase but too large a value of μ will cause instability. It is experimentally determined that the maximum value of the step size that can be used in the FXLMS algorithm is approximately:

$$\mu_{\max} = \frac{1}{p_{x'}(L+\Delta)} \tag{6}$$

where $p_{x'} = E[x'^2(n)]$ is the mean-square value or power of the filtered reference signal x'(n), L is the number of weights and Δ is the number of samples corresponding to the overall delay in the Secondary Path.

After convergence, the filter weights W(n) vary randomly around the optimal solution W_{opt} . This is caused by broadband disturbances, like measurement noise and impulse noise on the error signal. These disturbances cause a change of the estimated gradient $\nabla \hat{\xi}(n)$, because it is based only on the instantaneous error. This results in an average increase of the MSE, that is called the excess MSE, defined as, [9]

$$\boldsymbol{\xi}_{excess} = E[\hat{\boldsymbol{\xi}}(n)] - \boldsymbol{\xi}_{\min} \tag{7}$$

This excess MSE is directly proportional to step size. It can be concluded that there is a design trade-off between the convergence performance and the steady-state performance. A larger value of μ gives faster convergence but gives bigger excess MSE and vice-versa. Another factor that influences the excess MSE is the number of weights.

4 Simulation and Results

MATLAB platform is chosen for simulation purpose. The room impulse response is measured for an enclosure of dimension $12 \times 12 \times 8$ ft³ using an ordinary loudspeaker

and an uni-dynamic microphone kept at a distance of 1 ft which has been sampled at 8 KHZ. The channel response is obtained by applying a stationary Gaussian stochastic signal with zero mean and unit variance as input on the measured impulse response. Fig. 1 shows the MATLAB simulation of noise cancellation.

Fig. 1 (a) is the original noise, (b) is the noise observed at the headset. Fig. 1(c) is the inverse noise generated by an adaptive algorithm using the DSK, and Fig. 1(d) is the error signal between the noise observed at the headset and the inverse noise generated by the DSK. Fig. 1(d) shows that the error signal is initially very high, but as the algorithm converges, this error tends to zero thereby effectively cancelling any noise generated by the noise source.



The primary and secondary path impulse response with respect to time is presented in Fig. 2 and Fig. 3 respectively. The first task in active noise control is to estimate the impulse response of the secondary propagation path. This is usually performed prior to noise control using a random signal played through the output loudspeaker while the unwanted noise is not present. Fig.4 shows the true estimation of secondary path impulse response with the variation in coefficient value for the true, estimated and error signal with respect to time. The performance of FXLMS algorithm for ANC can be judged from the discussed results pertaining to an adaptive framework.



Fig. 2. Primary path impulse response



Fig. 3. True secondary path impulse response



Fig. 4. True estimation of Secondary path impulse response

5 Conclusion

FXLMS Algorithm is widely used in acoustic noise cancellation environment due to its simple real time implementation & low computational complexity. Although the algorithm has slow convergence but various modifications in the existing circuit can be done to improve the convergence. FXLMS algorithm has greater stability than other algorithms used in case of acoustic noise cancellation as shown in the paper. There is a wide range of modification scopes available in case of FXLMS algorithm which can improve the convergence. FXLMS algorithm can also be used in very high noise environment.

References

- 1. Elliot, S.J.: Signal Processing for Active Control. Academic Press, London (2001)
- Elliot, S.J., Nelson, P.A.: Active noise control. IEEE Signal Processing Magazine 10, 12– 35 (1993)
- 3. Lueg, P.: Process of silencing sound oscillations. U.S. Patent, 2043416 (1936)
- 4. Kuo, S.M.: Morgan, D.R.: Active Noise Control Systems-Algorithms and DSP Implementations. Wiley (1996)
- Kuo, S.M., Morgan, D.R.: Active Noise Control: A tutorial review. Proceedings of IEEE 87, 943–973 (1999)
- 6. Widrow, B., Stearns, S.D.: Adaptive Signal Processing. Prentice Hall, New Jersey (1985)
- Morgan, D.R.: An analysis of multiple correlation cancellation loops with a filter in the auxiliary path. IEEE Transactions in Acoustics, Speech, and Signal Processing 28, 454– 467 (1980)
- Boucher, C.C., Elliot, S.J., Nelson, P.A.: Effects of errors in the plant model on the performance of algorithms for adaptive feedforward control. IEE Proceedings F138(4), 313–319 (1991)
- Widrow, B., McCool, J.M., Larimore, M., Johnson, C.R.: Stationary and non-stationary learning characteristics of the LMS adaptive filter. IEEE Proceedings 64(8), 1151–1162 (1976)
- Butterweck, H.: A wave theory of long adaptive filters. IEEE Transactions on Circuits and Systems I:Fundamental Theory and Applications 48(6), 739–747 (2001)

- 11. Warnaka, G.E., Poole, L.A., Tichy, J.: Active acoustic attenuators. U.S. Patent 4473906 (1984)
- Elliott, S.J., Stothers, I.M., Nelson, P.A.: A multiple error LMS algorithm and its applications to active control of sound and vibration. IEEE Transactions on Acoustic, Speech and Signal Processing Processing 35, 1423–1434 (1987)
- 13. Rupp, M., Sayed, A.H.: Two variants of the FxLMS algorithm. In: IEEE ASSP Workshop on Applications of Signal Processing to Audio and Acoustics, pp. 123–126 (1995)
- 14. Rupp, M., Sayed, A.H.: Robust FxLMS algorithms with improved convergence performance. IEEE Transactions on Speech and Audio Processing 6(1), 78–85 (1998)
- Davari, P., Hassanpour, H.: A variable step-size FxLMS algorithm for feedforward active noise control systems based on a new online secondary path modelling technique. In: IEEE/ACS International Conference on Computer Systems and Applications, pp. 74–81 (2008)
- Kunchakoori, N., Routray, A., Das, D.: An energy function based fuzzy variable step size fxlms algorithm for active noise control. In: IEEE Region 10 and the Third International Conference on Industrial and Information Systems, pp. 1–7 (2008)
- Eriksson, L., Allie, M., Melton, D., Popovich, S., Laak, T.: Fully adaptive generalized recursive control system for active acoustic attenuation. In: IEEE International Conference on Acoustics, Speech, and Signal Processing, vol. 2, pp. II/253–II/256 (1994)

A Unique Low Complexity Parameter Independent Adaptive Design for Echo Reduction

Pranab Das¹, Abhishek Deb¹, Asutosh Kar¹, and Mahesh Chandra²

¹ Department. of Electronics and Telecommunication Engineering Indian Institute of Information Technology, Bhubaneswar, India ² Department of Electronics and Communication Engineering Birla Institute of Technology, Mesra, India {b210017, b210001, asutosh}@iiit-bh.ac.in, shrotriya@bitmesra.ac.in

Abstract. Acoustic echo is one of the most important issues in full duplex communication. The original speech signal is distorted due to echo. For this adaptive filtering is used for echo suppression. In this paper our objective is to cancel out the acoustic echo in a sparse transmission channel. For this purpose many algorithms have been developed over the period of time, such as Least Mean Square (LMS), Normalized LMS (NLMS), Proportionate NLMS (PNLMS) and Improved PNLMS (IPNLMS) algorithm. Of all these algorithms we carry out a comparative analysis based on various performance parameters such as Echo Return Loss Enhancement, Mean Square Error and Normalized Projection Misalignment and find that for the sparse transmission channel all these algorithm are inefficient. Hence we propose a new algorithm modified - μ - PNLMS, which has the fastest steady state convergence and is the most stable among all the existing algorithms, this we show based on the simulation results obtained.

Keywords: Acoustic Echo, Adaptive Filter, Echo Return Loss Enhancement, LMS, Mean Square Error, Sparse Transmission Channel.

1 Introduction

Echo is a delayed and distorted version of the original signal. Acoustic echo is mainly present in mobile phones, Hands free phones, Teleconference or Hearing aid systems which is caused due to the coupling of microphone & loudspeaker. Echo largely depends on two parameters: amplitude & time delay of reflected waves. Usually we consider echoes with appreciable amplitude & larger delays of above 1 ms, but in certain cases if the generated echo is above 20 ms then it becomes a major issue and needs to be cancelled. Thus, developers are using the concepts of Digital Signal Processing for echo cancellation to stop the undesired feedback and allow successful full duplex communication. [1]

To achieve echo free systems numerous methods have been proposed like echo absorbers, echo barriers, echo cancellers to name a few. But considering the

advancement of signal processing echo cancellation can be best done by the help of adaptive filtering. Adaptive filters are widely used due to its stability and wide scope for improvements. Adaptive filters can also be applied to low frequency noise. Echo can be suppressed in the most effective manner by continuous adaptation of the weights of the adaptive filter till the echo is completely cancelled out. System identification is generally used to generate a replica of the echo that is subtracted from the received signal. The echo canceller should have a fast convergence speed so that it can identify and track rapidly the changes in the unknown echo path. The convergence rate depends on the adaptive algorithm as well as the structure of adaptive filter used in AEC. In AEC, a signal is generated which is correlated to the echo signal but opposite in phase. By adding both the signals that is the signal corrupted by noise and the generated signal which is correlated and opposite phase to the actual echo signal, original signal can be made echo free. The generation of the signal is controlled by an adaptive algorithm which adaptively changes the weight of the filter used. AEC is classified into two categories i.e. "feed-forward AEC" and "feedback AEC". In feedback AEC method we use a controller to modify the response of the system. The controller may be like an addition of artificial damping. But in feed-forward AEC method a controller is used to adaptively calculate the signal that cancels the noise. In this paper "feed-forward AEC" approach is implemented to cancel the echo. This is because the amount of echo reduction achieved by feed-forward AEC system is more than that of feedback AEC system [2].

Further the nature of the transmission channel in AEC being sparse i.e. only few weight coefficients are active and all others are zero or tending to zero. The basic fundamental algorithms like LMS and NLMS suffer from slow convergence speed in these sparse channels. Hence a new modified algorithm was developed by Duttweiler named Proportionate NLMS for the sparse transmission channel systems [3]. The concept behind this is to assign each coefficient a step size proportionate to its estimated magnitude and to update each coefficient of the adaptive filter independently. But the main issue with the algorithm is that the convergence speed is reduced excessively after the initial fast period. A new kind of adaptive filtering process μ PNLMS was proposed to encounter the aforesaid problem. Here, the logarithm of the magnitude is used as the step gain of each coefficient instead of magnitude only. So this algorithm can consistently converge over a long period of time.

In this paper, we propose an algorithm to improve the performance of the MPNLMS algorithm to a greater extent in the sparse channel. The proposed algorithm adaptively detects the channel's sparseness and changes the step size parameter to improve the ERLE, Mean Square Error (MSE) and Normalized Projection Misalignment (NPM).

2 Existing Algorithms for AEC

A. LMS & NLMS:

It is one of the most widely used algorithms which has a weak convergence but is easy to use and is stable. It has two inputs one of which is the reference noise that is related with the noise that exits in the distorted input signal. The weight update equation for the LMS algorithm is given by:

$$\hat{h}(n+1) = \hat{h}(n) + 2x(n+1)e(n+1)$$
(1)

In case of LMS algorithm if the step size is too small then the adaptive filter will take too much time to converge, and if it is too large the adaptive filter becomes unstable and its output diverges .So it fails to perform well in echo cancellation. The recursive formula for Normalized Least mean Square (NLMS) algorithm is

$$\hat{h}(n+1) = \hat{h}(n) + 2\mu \frac{x(n+1)e(n+1)}{\|x(n+1)\|_2^2 + \delta_{NLMS}}$$
(2)

Here δ_{NLMS} is the variance of the input signal x(n) which prevents division by zero during initialization stage when x(n) = 0. Further to maintain stability the step

size should be in the range
$$0 < \mu < 2 \frac{E\{|x(n+1)|^2\}D(n+1)\}}{E\{|e(n+1)|^2\}}$$

Here $E\{|x(n)|^2\}$ is the power of input signal, $E\{|e(n)|^2\}$ is the power of error signal and D(n) is the mean square deviation. The NLMS algorithm though being efficient does not take into consideration sparse impulse response caused to bulk delays in the path and hence needs to adapt a relatively long filter. Also unavoidable noise adaptation occurs at the inactive region of the filter. To avoid this we need to use sparse algorithms, where adaptive step size are calculated from the last estimate of the filter coefficients in such a way that the step size is proportional to step size of filter coefficients. So active coefficients converge faster than non-active ones and overall convergence time gets reduced. [4]

B. PNLMS

In order to track the sparseness measure faster PNLMS algorithm was developed from NLMS equation. Here the filter coefficient update equation is different from the NLMS in having a step size update matrix Q, with rest of the terms remaining same and is given below as:

$$\hat{h}(n+1) = \hat{h}(n) + \mu \frac{Q(n)x(n+1)e(n+1)}{\|x(n+1)\|_2^2 + \delta_{PNLMS}}$$
(3)

Here, $\delta_{PNLMS} = \delta_{NLMS} / L$, and the diagonal matrix Q (n) is given by,

 $Q(n) = diag\{q_0(n), q_1(n), \dots, q_{l-1}(n)\}$

Here $q_I(n)$ is the control matrix and is given by,

$$q_{l}(n) = \frac{k_{l}(n)}{\frac{1}{L}\sum_{i=0}^{L-1}k_{i}(n)}$$
(4)

Also $k_l(n) = \max\{\rho \times \max\{\gamma, |\hat{h}_0(n)|, ..., |\hat{h}_{l-1}(n)|\}, |\hat{h}_l(n)|\}$

Here ρ and λ have values 5/L and 0.01 respectively. ρ prevents coefficients from stalling when they are smaller than the largest coefficient and also prevents

 $\hat{h}_{l}(n)$ from stalling during the initialization stage. [5]

C. IPNLMS:

It is an improvement over the PNLMS algorithm and employs a combination of proportionate (PNLMS) and non-proportionate (NLMS) updating technique, with the relative significance of each controlled by a factor α . Thus we have value of $k_I(n)$

$$k_{l}(n) = \frac{1 - \alpha}{2L} + (1 + \alpha) \frac{|h_{l}(n)|}{2 ||h(n)||_{1} + \varepsilon}$$
(5)

here ε is a small positive constant .Also results so that good choice for α are 0.-0.5,-0.75. The regularization parameter should be taken such so that same steady state misalignment is achieved compared to that of NLMS using same step size. [6] We have

$$\delta_{IPNLMS} = \frac{1 - \alpha}{2L} \delta_{NLMS} \tag{6}$$

D. MPNLMS:

This algorithm is an efficient one for a sparse transmission channel and has a steady convergence over a period of time. Unlike the previously discussed proportionate algorithms which have a faster convergence during the initial period but slows down over a period of time. In this algorithm we calculate the step size proportionate to the logarithmic magnitude of the filter coefficients. [7]

3 Proposed Algorithm

Among all the algorithms discussed above for a sparse channel, which is our transmission channel of interest, the μ -PNLMS algorithm has the fastest convergence. The existing classical NLMS algorithm's filter weight coefficients converge slowly in such a channel. Hence PNLMS and IPNLMS algorithms were designed specifically for echo cancellation in sparse transmission channels, but after

as

initial fast convergence of filter weight vectors, these algorithms too fail.[6]-[7] So we propose μ -law PNLMS algorithm for a sparse channel. Here instead of using filter weight magnitudes directly, its logarithm is used as step gain of each coefficient. Hence μ -PNLMS algorithm can converge to a steady state effectively for a sparse channel. This algorithm calculates the optimal proportionate weight size in order to achieve fastest convergence during the whole adaptation process until the adaptive filter reaches its steady state.

Implementation:

Step 1: The weight update equation for the proposed algorithm is given as:

$$\hat{\mathbf{h}}(n+1) = \hat{h}(n) + \mu \frac{Q(n)x(n+1)e(n+1)}{\|x(n+1)\|_2^2 + \delta_{MPNLMS}}$$
(7)

where;

Here,

$$\delta_{MPNLMS} = \delta_{NLMS} / L \tag{8}$$

Step 2: The control matrix $q_l(n)$ which is the matrix containing the diagonal elements of the Q(n) is given as,

$$q_{l}(n) = \frac{k_{l}(n)}{\frac{1}{L}\sum_{i=0}^{L-1}k_{i}(n)}$$
(9)

Step 3: Finally the value of $k_l(n)$, which is the main differentiating factor in case of the proposed algorithm is given as:

$$k_{l}(n) = \max\{\rho \times \max\{\gamma, F(|\hat{h}_{0}(n)|)...,F(|\hat{h}_{l-1}(n)|)\}, F(|\hat{h}_{l}(n)|)\}$$
(10)

 $F(|\hat{\mathbf{h}}_{l}(n)|) = \frac{\ln(1+\mu|\hat{h}_{l}(n)|)}{\ln(1+\mu)},$ (11)

 $|\hat{\mathbf{h}}_{L}(n)| \le 1$ and $\mu = 1/\varepsilon$. Constant 1 is taken inside the algorithm to avoid infinity value when the value of $|\hat{\mathbf{h}}_{l}(n)| = 0$ initially. The denominator $\ln(1 + \mu)$ normalizes $F(|\hat{\mathbf{h}}_{L}(n)|)$ in the range [0, 1]. The value of ε should be chosen based on the noise level and usually it is taken as 0.001.

4 Results and Discussion

MATLAB platform is chosen for simulation purpose. The room impulse response is measured for an enclosure of dimension $12 \times 12 \times 8$ ft³ using an ordinary loudspeaker and an uni-dynamic microphone kept at a distance of 1 ft. which has been sampled at 8 KHZ. The channel response is obtained by applying a stationary Gaussian stochastic signal with zero mean and unit variance as input on the measured impulse response.



Fig. 1. AEC using proposed Algorithm

Fig. 1 shows the MATLAB simulation of acoustic echo cancellation using the proposed algorithm. Fig.1 has three separate graphs where the first is the original noise present in the system ,the second one is the graph of original signal along with the noise signal and the third graph is the original signal recovered using the proposed algorithm.



Fig. 2. ERLE Comparison of existing and proposed algorithms



Fig. 3. NPM comparison of existing and proposed algorithms

In Fig. 2 we show the ERLE comparison of PNLMS, IPNLMS, μ – PNLMS and the proposed algorithm, the result clearly shows that the ERLE for the proposed algorithm is maximum. Further in Fig. 3 we show the comparative analysis of NPM of PNLMS, IPNLMS, μ – PNLMS and the proposed algorithm which clearly indicates that has the least value for our proposed algorithm.

The analysis is carried out by taking a signal x(n) as an input to the various adaptive algorithms and the desired signal is taken as d(n). The error signal thus generated is minimized continuously in an adaptive manner by setting the step size to an optimum value. Our aim is to obtain the quickest steady state convergence for minimizing the error or the acoustic echo, which is obtained using our proposed algorithm. This can be shown through the comparison of ERLE and NPM for all the algorithms including our proposed algorithm. And based on our simulation results we obtain that our proposed algorithm has the maximum ERLE and the least NPM value among all the algorithms such as PNLMS, μ – PNLMS and IPNLMS.

5 Conclusion

In this paper we do a detailed study of the existing algorithms for acoustic echo cancellation and find out that only proportionate algorithms are capable of acoustic echo cancellation in sparse channel. But among the existing proportionate algorithms PNLMS, IPNLMS though have initial fast convergence they fail to adapt as time progresses [7]. And only μ – PNLMS algorithm has a good steady state convergence. Now to further improve the performance of μ – PNLMS algorithm we propose our own algorithm which we have shown to have the most efficient steady state convergence for transmission channels having sparse impulse response. This we ascertain based on our simulation results, where we take into consideration various performance measurement parameters like ERLE and NPM.We have obtained that our algorithm gives the maximum ERLE amongst all the existing algorithm and also the least NPM value. So we can safely conclude that our proposed algorithm is the most efficient one for acoustic echo cancellation in a sparse medium.

References

- 1. Verhoeckx, N.A.M.: Digital echo cancellation for base-band data transmission. IEEE Trans. Acoustic, Speech, Signal Processing 27(6), 768–781 (1979)
- 2. Haykin, S.: Adaptive Filter Theory, 2nd edn. Prentice-Hall Inc., New Jersey (1991)
- Khong, A.W.H., Naylor, P.A., Benesty, J.: A low delay and fast converging improved proportionate algorithm for sparse system identification. EURASIP Journal of Audio Speech Music Processing 2007(1) (2007)
- Diniz, P.S.R.: Adaptive Filtering, Algorithms and Practical Implementation. Kluwer Academic Publishers, Boston (1997)

- Wang, X., Shen, T., Wang, W.: An Approach for Echo Cancellation System Based on Improved NLMS Algorithm. School of Information Science and Technology, pp. 2853– 2856. Beijing Institute of Technology, China (2007)
- Paleologu, C., Benesty, J., Ciochin, S.: An Improved Proportionate NLMS Algorithm based on the Norm. In: International Conference on Acoustics, Speech and Signal Processing, pp. 309–312 (2010)
- Deng, H., Doroslovachi, M.: Proportionate adaptive algorithms for network echo cancellation. IEEE Trans. Signal Processing 54(5), 1794–1803 (2006)

On the Dissimilarity of Orthogonal Least Squares and Orthogonal Matching Pursuit Compressive Sensing Reconstruction

Arvinder Kaur and Sumit Budhiraja

Department of Electronics and Communications Engineering, UIET, Panjab University, Chandigarh, India sharma.arri@gmail.com, sumit@pu.ac.in

Abstract. Compressive sensing is a recent technique in the field of signal processing that aims to recover signals or images from half samples that were used by Shannon Nyquist theorem of reconstruction. For recovery using compressed sensing, two well known greedy algorithms are used- Orthogonal matching pursuit and orthogonal least squares. Generally these two algorithms are taken as same by the researchers which is not true. There is a remarkable difference between the two algorithms that is pointed out in this paper with the simulation results. The previous article clarifying the difference between these two algorithms are emphasized on theoretical difference and does not show any reconstruction simulation difference with these two algorithms and reason to preference over basis pursuit method . The key aim of this paper is to remove the confusion between the two algorithms on the basis of theory and reconstruction time taken with the output PSNR.

Keywords: Compressive sensing, Reconstruction, Orthogonal least squares, orthogonal matching pursuit.

1 Introduction

Compressed sensing is an innovative theory in the field of signal processing that aims to reconstruct signals and images from what was previously treated as incomplete information. It uses relatively few measurements than those were needed in the Shannon Nyquist theory of signal reconstruction. This theory comes from the fact that most of the signal information is carried by only few of coefficients and all other are get wasted while reconstructing that signal [1]. Then it gives a way to collect only those components in the first stage that adds up in signal recovery. The practical need of this study comes with the shortage of storage space for increasing data transferred from here to there.

Compressed sensing has been used in many fields like medical imaging, Radar, Astronomy, speech processing where the task is to reconstruct a signal. For example, in medical imaging CT (computerized Tomography), by exposing x-rays one can generate image of inside of human body. For the complete scanning process, patient is exposed to radiations for a large span of time, which are harmful for him. By using compressive sensing, only the desired samples are to be taken, so scanning time reduces to a great extent by decreasing the number of samples to be taken. The mechanism of reconstruction using sparse matrix is shown in Fig. 1.



Fig. 1. Compressed sensing framework

Where Y is the reconstructed signal and A' is the random measurement matrix. Gaussian and Bernoulli matrices are mostly used as random measurement matrices. For Gaussian matrices the elements are chosen randomly as independent and identically distributed. These elements can have the variance as from [3]

Variance=1/K

For the Bernoulli matrices [3] with equal probability each element takes value of $1/\sqrt{K}$ or $+1/\sqrt{K}$

For reconstruction using compressed sensing two type of algorithm are used basis pursuit and greedy algorithms. Basis pursuit algorithms are quite simpler to implement as compared to greedy pursuit. The greedy algorithms generally used are orthogonal matching pursuit and orthogonal least squares give faster recovery. Generally these two algorithms are taken as same but there is certain difference in them [8]. For the greedy algorithms, the projection on selected signal elements is orthogonal. OLS selects the entity that minimizes the residual norm after signal's projection onto the selected entities [4]. It has been proved by Thomas Blumensath and Mike E. Davies in [4]. This particular entry is the one which best approximates the current residual. This marks the difference of OLS from OMP. Generally OLS is little bit more complex than OMP, but due to the same output results given by two algorithms, both are taken as same by the researchers. There had been previous work on clarification between two algorithms, but most of it is based on theoretical concepts. In this paper the confusion on the similarity part of both algorithms is removed by showing the simulation results of both the algorithms for same results. The reconstruction time taken by both is also considered.

2 Orthogonal Matching Pursuit

Tropp and Gilbert had given an idea for orthogonal matching pursuit algorithm for reconstruction with less number of measurements [5-6]. It has been proved in the

literature by Huang and Rebollo that reconstruction time can be reduced to a great extent by using this particular algorithm [13], [15]. Suppose that the column vector of A are normalized such that

$$|| A_i ||_2 = 1$$
, for i=1,2,...k

A(x) be a subset of A for x C {1,2,..,k}. To start the algorithm, signal support is needed to be calculate from pseudo inverse A` of the measurement matrices A, as

P=(A) Y

Where A' is the complex conjugate of measurement matrices A and we know that

$$A^{=}(A^{*}A)^{-1}A^{*}$$

During the implementation a matching operation is performed between the matrix A entries and the calculated residuals of P [2, 14]. At the end of all iterations, the complete signal is generated. The implemented algorithm includes following steps

- 1. Initialize the residual of Y as $s_0 = y$
- 2. Initialize the set of entries that are to be selected as A(c) = null
- 3. Start an iteration counter. Let i=1
- 4. Generate an estimate of given signal per iteration.
- 5. Solve the maximization problem

max | A_j, s_{i-1} |

In which A_j is any variable.

- 6. To A(c), the set of selected variables, add A_i, and update c after every result.
- 7. It is to be noted that the algorithm does not minimize the residual error after its orthogonal projection to the values.
- 8. Update result after the projection. The entry selected have a minimum error as $r^n=S-S^n$
- 9. One coordinate for signal support for P is calculated at the end after computing the residual.
- 10. Set i=i+1, to end the iterations of algorithm.

3 Orthogonal Least Squares

On the contrary to matching pursuit algorithm, there is another method in the literature based on least squares [7], [9]. In OLS, after the projection into the orthogonal subspace, OLS selects the element with smallest angle [4]. This element best approximates the current residual. For the least squares method, a vector is needed to be calculated of few non-zero elements such that the squared error is small [4]. That's why the name least squares is there. The vector calculated must be with dimensions

$$P \in D_K and$$

 $A \in D^{Np * Ny}$

An approximation to the signal P is calculated at the end of an iteration and result is updated after every sequence. The complete procedure goes same for least squares algorithm [12], [14]. Initially a counter is started and a residual is assumed and residual is updated after every projection result. This algorithm selects the entry having a minimum error from the sub matrix [4]. Just after the orthogonal projection step, the algorithm minimizes the residual error. The approximation error is given by

$$e^n = P - P^n$$

After the projection onto the signal entries, calculate the maximization problem as

$$j_{max} = ||A - (A_i^* A_i)^{-1}A||_2$$

Solving A $_{i}^{-1}$ is an inverse problem here. The pseudo inverse A $_{i}^{-1}$ of the random measurement matrices is taken from

$$P_n = A_i^{-1} A$$

Thus the signal is reconstructed from the inverse problem solution [10].

4 Simulation Results and Analysis

The focus of this paper is to highlight difference between the two mentioned greedy algorithms. An estimate for matrix A is calculated and its columns are used to calculate the inner products required to select elements from signal. When the number of elements to be selected is quite smaller than the number of column vectors of matrix A; OLS is complex because all elements need to be orthogonalized [4].



Fig. 2. (a) Original randomly generated signal (b) original signal reconstructed using Orthogonal Least Squares



Fig. 3. (a) Original randomly generated signal (b) original signal reconstructed using Orthogonal Matching Pursuit

It has been proved previously that the greedy algorithms are faster than the basis pursuit compressive sensing reconstruction. This is shown in Fig.2(a),(b) and Fig.3(a),(b) A comparison of elapsed time for reconstruction and for PSNR for the output signal is shown in Table 1. A comparison of both the algorithms is done with Justin Romberg's Basis pursuit's algorithm [11].

Algorithm	Elapsed Time(s)	PSNR(db)	
OLS	0.037765	83.5091	
OMP	0.052453	83.5217	
BP	0.731953	81.7943	

Table 1. Comparison of elapsed time and PSNR for both the algorithms

In Table 1, the reconstruction time is given in seconds and is calculated from the beginning of the generation of sparse matrix up to the generation of output signal. A slight difference in PSNR is observed for the same signal recovery by using the two algorithms, which marks the difference between the two algorithms, i.e. proves that both are not same. The reconstruction results are obtained by implementing the greedy algorithms in a multi stage manner, with less number of dimensions. The PSNR obtained is higher than the L-norms or the basis pursuit method of compressive sensing reconstruction. It is clear from the table that greedy algorithms are faster than the basis pursuit algorithm.

5 Conclusions

The results from this paper demonstrate that both OLS and OMP algorithms gives same recovery results but there is a significant difference between the two in terms of